Gender Impacts
of Government
Revenue Collection:
The Case of
Taxation

Kathleen Barnett and Caren Grown



# Gender Impacts of Government Revenue Collection: The Case of Taxation

Kathleen Barnett Caren Grown



This report is part of the Commonwealth Economic Paper Series
published by the Economic Affairs Division of the Commonwealth Secretariat
Commonwealth Secretariat
Marlborough House, Pall Mall
London SW1Y 5HX, United Kingdom

© Commonwealth Secretariat, 2004

The authors can be contacted at the International Center for Research on Women 1717 Massachusetts Avenue, N.W., Suite 302, Washington, DC 20036

Views and opinions expressed in this paper are the responsibility of the authors and should in no way be attributed to the institutions to which they are affiliated.

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording or otherwise without the permission of the publisher.

Designed and published by the Commonwealth Secretariat Printed in Britain by Formara Ltd.

Wherever possible, the Commonwealth Secretariat uses paper sourced from sustainable forests or from sources that minimise a destructive impact on the environment.

ISBN 0-85092-788-9 Price: £10.00

Web site: http://www.thecommonwealth.org

## **Acknowledgements**

Thanks are due to Daniel Puskin and Zahia Khan, who provided excellent research assistance, and to Simel Esim, Amy Schwartz, Stephanie Seguino, Janet Stotsky, Irene Van Staveren and the members of the Commonwealth Secretariat Expert Review Panel, including Patricia Apps, Debbie Budlender, Diane Elson, Sue Himmelweit and Haroon Akram-Lodhi, who provided insightful comments on an earlier draft of this paper. We also gratefully acknowledge the support of Guy Hewitt, Donna St Hill and the Commonwealth Secretariat, who made this project possible. Any errors are our own.

# **Contents**

1	Introduction	1
2	Gender Analysis: Basic Concepts	3
2.1	The Unpaid Care Economy	3
2.2	Gender Differences in Employment	4
2.3	Gender Differences in Household Decision-making	5
2.4	Gender Differences in Property Rights	6
2.5	Gender and Other Social Stratifiers	6
2.6	Gender Analysis of Taxation	7
3	Taxation and Developing Countries	8
3.1	Taxation and Tax Policy Analysis	8
3.2	Tax Systems in Developing Countries: Background	9
3.3	Tax Systems in Developing Countries: Current Issues	9
3.4	Tax Terminology and Concepts	16
	3.4.1 Tax Base	16
	3.4.2 Tax Administration	16
	3.4.3 Tax Burden	16
	3.4.4 Tax Incidence	16
	3.4.5 Tax Incentives/Tax Expenditures	17
	3.4.6 Tax Reform	17
	3.4.7 Fiscal Decentralisation	18
4	Tax Policy Analysis: Concepts and Critiques	20
4.1	The Behavioural Response to Taxation	20
4.2	Equity	21
4.3	Efficiency	23
4.4	Ease of Tax Administration	24
4.5	The So-called 'Efficiency/Equity Trade-off'	24
5	Gender Impacts of Taxation: A Review of the Literature	26
5.1	Summary of Possible Gender Bias by Tax: Explicit and Implicit Bias	26
5.2	A Hypothetical Gender-Tax Typology	28
5.3	Personal Income Taxes	30

Bibli	ography	61
8	Conclusions and Recommendations	57
7.3	Summary	55
7.2	Methodologies and Tools Used in Gender-responsive Government Budgeting	55
7 2	7.1.4 Macrosimulation Modelling  Mathedalogies and Tools Head in Conden responsive Covernment	54 55
	7.1.3 Microsimulation Modelling	54
	7.1.2 Tax Incidence Analysis	53
	7.1.1 Tax Burden Analysis	53
7.1	Traditional Methodologies and Tools for Tax Policy Analysis	53
7	Methodological/Analytical Tools for Gender-revenue Analysis	53
6.4	Summary of Debates and Issues	52
6.3	Gender-specific Tax Policy Issues	51
6.2	General Issues of Tax Policy in Developing Countries	50
6.1	Tax Policy Debates	50
6	Tax Policy Debates and Issues	50
5.10	User Charges	47
	Tax Reform and Gender	45
	5.8.1 Gender Bias in Trade Taxes	45
5.8	Trade Taxes	44
	5.7.1 Gender and Tobin Taxes	44
5.7	The Tobin Tax Proposal	43
	5.6.1 Gender Bias in Corporate Income Taxes	42
5.6	Corporate Income Taxes	41
	5.5.1 Gender Bias in Commodity Taxes	39
5.5	Commodity Taxes (VAT, Sales, Excise)	38
	5.4.1 Gender Bias in Payroll Taxes	37
5.4	Payroll Taxes	36
	5.3.6 Impact of PIT on Fertility	35
	5.3.5 Impact of PIT on Time Use and Household Production	34
	5.3.4 Impact of PIT on Married Women's Labour Supply	33
	5.3.3 Gender Implications of PIT Joint Filing	32
	5.3.1 Gender Bias in Personal Income Taxes 5.3.2 Gender Implications of PIT Individual Filing	30 32
	5.3.1 Gender Bias in Personal Income Taxes	30

## 1

## Introduction

Efforts to integrate a gender perspective into public budgeting decisions have been going on for almost 20 years. In 1984, the Australian government introduced the first 'gender budget' exercise, which led ultimately to a gender review of all federal, state and territorial government expenditures and some elements of revenue. At the 1995 United Nations World Conference on Women in Beijing, governments made a commitment to incorporate a gender perspective into budgetary processes in order to support gender equality. Since then, work by women's NGOs, other civil society organisations, academics and multilateral organisations has shown that the analysis of government expenditures can be an important tool for addressing gender inequities.¹ Analysts and activists are now increasingly interested in how to bring a gender perspective to the revenue side of the budget, as well as how to use gender revenue analysis as a tool for advancing gender equality.

Taxation and expenditure need to be analysed together for a full understanding of the income and gender impacts of government fiscal policy. The level of taxes to be levied is decided in conjunction with the level of needed expenditures. The ultimate ratio of tax burden/social benefit for individual women and men depends on the gender distribution of both taxes paid and services received, as well as the gender distribution of the long-term gains from social and economic development. The goal of gender revenue analysis is to identify and monitor the flow of sufficient financial resources so that gender equity is achieved in revenue generation and women and men, and girls and boys, benefit equally from programmes and services.

This paper reviews the literature on the gender dimensions of taxation and the implications for tax policy with special reference to developing countries. Taxation is a critical subject for a gender analysis of development policy for several reasons. First, in the developing world, the majority of the population – and the vast majority of women – are poor, so adequate financing of public services is a pressing issue with special gender relevance. Second, since taxes are governments' principal own-source revenues, tax policy is at the heart of the public debate on what services government should provide and who should pay for them. Third, taxes in developing countries constitute 10–40 per cent of a country's gross domestic product (GDP), which means that out of every rupee or pound earned, a significant share goes to the government (IMF, 2002). Such a large extraction of resources affects all aspects of social and economic life and – matched with expenditures – determines the path and distribution of development.

The Commonwealth Secretariat commissioned this paper as part of its commit-

ment to integrate gender concerns into economic policy. The paper follows important prior work on the gender dimensions of public expenditure. Its purpose is to provide information to assist in the analysis of potential gender bias in tax systems and the design of gender-sensitive revenue measures.

The intended audience is a broad spectrum of government, NGO and civil society organisations, as well as economists concerned with public finance, macroeconomics and gender. The paper has an applied rather than an academic focus, and the material and language are designed to be accessible to readers without a formal public finance background. Since the issue of gender bias in taxation is a fairly new discussion, the literature on the subject is limited. There is a large literature on gender biases in the personal income tax systems of developed countries and a relatively smaller literature on the gender dimensions of direct and indirect taxation in developing economies. The literature from developed countries, although specific to conditions of high-income countries, provides helpful information on how gender bias can be identified and corrected in current tax systems, and avoided in future tax policy decisions.

The structure of the paper is as follows. Section 2 explains the concept of gender, and the implications of gender for taxation policy. Section 3 provides an overview of taxation issues in developing countries. Section 4 outlines some basic concepts of tax analysis and provides a feminist critique of them. Section 5 reviews the existing literature on the gender dimensions of tax policy, including a general overview of types of taxes and specific literature reviews on the gender impact of taxation on labour supply, household production and time use, and fertility. It also reviews the gender impacts of user charges which, while not taxes, are also a source of government revenue. Section 6 summarises current tax policy debates and issues, and Section 7 discusses methodologies and tools for gender revenue analysis, and data and modelling requirements. Section 8 concludes with recommendations on how to improve gender equity in tax policy and suggests areas for future research.

# **Gender Analysis: Basic Concepts**

Gender analysis suggests four issues or 'stylised facts' that are useful for understanding the impact of taxation on men and women.<sup>2</sup> These are the importance of:

- (a) women's work in the unpaid care economy;
- (b) gender differences in paid employment, including formal/informal employment, wages and occupational segregation;
- (c) gender differences in household decision-making regarding consumption, savings and investment;
- (d) gender differences in property rights.

## 2.1 The Unpaid Care Economy

In every country around the world, women have primary responsibility for caring for children and dependents. Women do most of the work in the unpaid economy (also called the domestic, social reproduction or reproductive economy). There are three types of unpaid work. Reproductive work consists of managing a household, cooking, cleaning, gathering fuel and hauling water, maintaining the home in good condition, and caring for family members, friends and neighbours. As Diane Elson (1996) points out, these are all vital services for the paid economy. In many developing countries unpaid work also includes subsistence production – production for home use of goods and services that in principle could be marketed, such as food, clothing and other items. Finally, unpaid work includes unpaid community work: for example soup kitchens organised by women in poor neighbourhoods, groups of mothers organising care of children, the elderly or those who are sick or disabled, and (unpaid) work for local or national non-profit charitable organisations.

Unpaid care work is not counted in most systems of national accounts, although efforts are underway in about 20 countries to collect satellite accounts.<sup>3</sup> Time use surveys from around the world show that women spend long hours in work that is not paid. Females work longer hours than males in nearly every country – the difference ranges from as little as eight minutes a day in Republic of Korea to two hours in mountainous regions of Nepal to three hours in rural Kenya (United Nations, 2000). The presence of small children (under five years old) increases unpaid work substantially more for women than for men.<sup>4</sup>

The gendered nature of unpaid work affects women throughout the life cycle – from girlhood through older age – and has enduring consequences (Gornick, 1999).

Recent research has begun to identify and quantify the 'costs' to women of their unpaid labour (Folbre, 1994; Davies and Joshi, 1994) and to explore the implications for taxation and benefits policies.

## 2.2 Gender Differences in Employment

Women's labour force participation rates are lower than men's, although women tend to have a greater combined labour burden consisting of paid and unpaid activities in the world economy (UNDP, 1995). Within paid employment, women are also disadvantaged relative to men. Worldwide, women are concentrated in clerical, sales and service jobs traditionally regarded as 'female' and are under-represented in production, transport, administrative and managerial jobs, in which men predominate (Mehra and Gammage, 1999). Even within sectors, men and women are found in different jobs; for instance, although both men and women are teachers in many countries, men are more likely to be concentrated in secondary or higher education, while women tend to be found in nursery or primary education (Anker, 1998). Moreover, women enter and exit the labour force more frequently, which means their participation is more discontinuous than that of men, and they are more likely to be in part-time and seasonal jobs, while men have full-time positions.

Women earn less than men, even after controlling for standard human capital variables (age and education). Gender gaps in earnings are among the most persistent forms of inequality, although they have narrowed in some countries over the last decade (Tzannatos, 1999; Artecona and Cunningham, 2002; Oostendorp, 2002). In other countries, however, wage gaps have widened (Standing, 1989, 1999; Mehra and Gammage, 1999). Even in the East Asian countries which have grown rapidly, relying on exports produced by female labour, gender wage gaps remain persistently large and have worsened in some cases (Seguino, 2000).

In developing and transitional economies, the bulk of new employment has been in the informal economy. Informal employment is market-oriented employment in small workshops and family businesses, subcontracted work undertaken in the home ('homeworking') and domestic work for others. It is work without secure contracts, worker benefits or social protection. It includes three types of occupational status – employee and self-employed (both of which are paid), and unpaid family worker, where the worker does not herself receive money but the family member (often the male household head) directing the business does receive payment from customers.

Informal employment is a larger source of employment for women than for men. Women constitute 84 per cent of the informally employed in sub-Saharan Africa (compared to 63 per cent of men) and 58 per cent of the informally employed in Asia and Latin America (compared to 48 per cent of men). Within informal employment, women tend to concentrate in street vending, home-based work and as industrial outsourcers who work at home (ILO, 2002).

The result of women's employment profile – their lower relative earnings and their predominance in informal employment – is that they are unlikely to bear a large share of the personal income or direct tax burden. However, indirect taxes, which are a heavier burden on the poor, are likely to have a greater impact on women than on men.

## 2.3 Gender Differences in Household Decision-making

Not only do women allocate their time differently from the way men do – between paid market work, household production and leisure – but there is also evidence that gender relations and bargaining power among household members affect the types of expenditures households make, the amount and type of savings, and other allocation decisions.

Across a wide range of cultures, empirical studies have revealed gender differentials in consumption (Haddad *et al.*, 1997). Women, compared to men, tend to spend a higher proportion of income under their control on goods such as food, education and healthcare that enhance the wellbeing and capabilities of children (Guyer, 1988; Thomas, 1992). Changes in the relative prices of these items are likely to affect women's expenditures and time allocations. For instance, a rise in food prices may have several effects: women may reduce their purchases of food and spend more time processing food at home. Or they may cut back on consumption, substituting cheaper (and perhaps less nutritious) foods in their diets. Taxation that changes the relative prices of goods may affect women's consumption decisions.

Women and men may also differ in their savings and investment behaviour. Floro and Seguino (2002) review the evidence from developed and developing countries and conclude that significant gender differences exist in individual retirement savings and investment decisions. In the United States, for example, a study by Sunden and Surette (1998) shows that gender and marital status influence investment allocation decisions; research from European countries echoes this finding. In developing countries, some evidence has shown that women and men may have different motives for savings, although evidence on the gender differences in savings behaviour is inconclusive.

There has been much research on the household decision-making process that underlies choices about consumption, savings and investment. This process is characterised both by co-operation and conflict and is influenced by the relative bargaining power of adults within the household.<sup>5</sup> Folbre (1997) and others argue that a household member's bargaining power depends on two factors: a) material (economic) factors internal to the household; and b) factors external to the household that influence its wellbeing. Material factors include owned assets, education, kinship, wages and employment. External factors include belief systems, tax and transfer policies, and political and legal structures such as property rights and divorce laws. The latter factors affect positions in household bargaining since they mediate the actual power that

material resources will confer on an individual in the household. Relative changes in any of the factors that affect an individual's bargaining power exert an influence on the allocation of household resources among alternative uses.

In sum, the literature suggests that it is important to look within households and assess the impact of taxes from the point of view of individual women and men, recognising that they have different streams of income, different paid and unpaid work responsibilities, and different propensities for consumption, savings and investment.

## 2.4 Gender Differences in Property Rights

In many developing countries women are frequently denied the right to own and inherit property. In many regions of Africa men hold formal land title. Social norms may dictate that businesses and other productive assets are owned by male family members although women may supply labour to family businesses. The gender asset gap in land is also significant in South Asia and Latin America, where women rarely represent more than one-quarter of landowners (Agarwal, 1995; Deere and Leon, 2003). Developed countries also have a legacy of unequal property rights. Women's lack of control over 'immovable property', i.e. land and houses, is mirrored by gender-based inequities in ownership of 'movable' property, such as businesses, equipment, furniture, clothing and personal items, household goods and capital. These patterns place women at a strong disadvantage in terms of securing a place to live and maintaining resources for their livelihoods.

In many developing countries, including those in South Asia, sub-Saharan Africa and Latin America, laws, policies and/or customary practices restrict women's ability to own or inherit property and reduce their capacity to manage risk successfully. Although women may not be taxed currently as property owners, as countries develop and as women acquire property rights in land and housing, they will become a more important part of the tax net.

#### 2.5 Gender and Other Social Stratifiers

It should be noted that gender analysis does not only entail analysing women as a distinct group in relation to men as a distinct group. Gender is a social stratifier that interacts with other important social stratifiers such as class, race, ethnicity and location (e.g. being rural or urban). Smith (2000), in his analysis of the impact of taxation on women in South Africa, explains why this additional disaggregation is important. Of women who are economically active, 28 per cent are unemployed, compared to 19 per cent of men, but of black African women, 35 per cent are unemployed compared to 25 per cent of black African men. The comparable unemployment rates for white women and men are 6 per cent and 3 per cent respectively, and for Indian women and men, 13 per cent and 9 per cent respectively. Thus, especially for analysing the incidence or

burden of taxation within a country, it is important to go beyond the 'average woman' to examine gender differences together with income and, if possible, with race, ethnicity and other important social markers.

## 2.6 Gender Analysis of Taxation

Few gender budget initiatives have examined the revenue side of the budget. Budlender (2000) suggests there are both political and technical challenges to be overcome in analysing revenue from a gender perspective. On the political side, she notes that gender-sensitive revenue analysis in South Africa encountered several political sensitivities because it came up against corporate and other wealthy interest groups, challenged moves toward decentralisation and raised fundamental questions about political strategies (such as privatisating public enterprises) to address budget deficits. More generally, she notes that gender-sensitive revenue analysis also encounters technical obstacles: analyses must be country specific because the structure of taxation and revenue differs greatly across countries and because of the different types (and even lack) of linkages between revenue and benefits. A second technical difficulty is that governments simply do not collect the sex-disaggregated data that are needed for tax incidence analysis. Where personal income taxes are collected on a joint (as opposed to an individual) basis, disentangling the gender effects can also be challenging.

Assessing taxation and revenue from a gender perspective, therefore, may not be an easy task. Nonetheless, as gender-responsive budgeting takes hold, scholars and practitioners are beginning to devise tools and methodologies to tackle a complex and difficult subject, an issue that is discussed in more detail in Section 7 of this paper. But first, Sections 3 and 4 provide some background information on taxation in developing countries, and set out the basic concepts and tools of tax analysis.

# **Taxation and Developing Countries**

## 3.1 Taxation and Tax Policy Analysis

In both developed and developing countries, governments collect taxes to fund public services. Although many developing countries are highly dependent on foreign aid, taxes are the principal own-revenue source. Other own-revenue sources are non-tax revenue, which includes fees, licences, mineral rights, etc., and capital revenue, which includes income from sales of government assets, including privatisations.<sup>8</sup>

Since taxes are the principal source of recurring revenue under government control, in all countries tax policy is at the heart of the political debate on the level of public services that should be provided and who should pay for them. The resolution of the debate varies by country and within countries over time. In democratic countries, the resolution depends on the ability of political parties, factions and interest groups to influence policy decisions and to influence voters to support these decisions. Groups without political or economic power, such as the poor and women, are often excluded from this debate and from tax policy decisions.

The goal of tax analysis is to identify the potential macro and micro impacts of tax policy options on individuals, businesses and economic growth, so that policymakers and the public have full knowledge of the impact of tax policy decisions. On the macro side, a gender analysis of tax policy would include analysis of the overall fiscal position of the budget, since deficits and options to close these can have explicit and implicit gender implications. Questions to be considered from a gender perspective might include: Are the estimates of revenue and expenditure, and of the projected budget surplus/deficit, reasonable?; What are the proposed policies for addressing government deficits and national debt?; What are the budget's expected effects on economic growth, inflation and employment? On the micro side, a gender analysis of tax policy would explore the implications for individual and household behaviour – in terms of labour supply, household production and time use, and marriage and fertility. Within both the macro and micro gender analysis, it is particularly important to address differential impacts by income, geography, ethnicity and race.

Tax analysis should also provide cost-benefit estimates of tax policy options, and ensure that the design of tax incentives (discussed in Section 3.4.5) is constructed to include provisions for proper implementation and periodic evaluation. Tax analysis, therefore, is essential to ensuring that the political debate regarding tax policy is fully informed of the social as well as fiscal impact of tax policy options, including the impact on income and gender equity.

## 3.2 Tax Systems in Developing Countries: Background

The tax system of each country reflects its specific history, legal tradition, political structure and economic base. Developing country tax systems originated from the traditions of their colonial powers, with little relationship to the actual conditions or interests of the country. Initially these systems were designed to extract wealth from the colony for the benefit of the coloniser.<sup>9</sup>

At the time of independence, most ex-colonies inherited tax systems based on early twentieth-century European conditions and a fiscal system in collapse – with few or no revenue sources, a weak civil service, and low levels of human and physical capital. In some developing countries, for example Guinea and Mozambique, the colonising power had consciously destroyed economic assets as it withdrew. In others, for example Angola and Guatemala, economic destruction had resulted from civil conflict sponsored by major economic powers or, as in Mozambique and Namibia, by apartheid South Africa.

Developing countries that adopted a market economy were encouraged by major commercial banks to borrow freely at low interest rates, and these liabilities turned into unsustainable deficits when interest rates turned high. Countries that followed a socialist path often suffered economic sanctions as well as political disruption by major powers, and/or stagnated economically because of misguided economic policies. Countries in transition from socialism, as well as post-apartheid South Africa, were faced with re-orienting their entire societies and economies to deliver to the majority rather than a minority, drafting new constitutions and constructing new government structures. Overcoming these historical conditions continues to be a major challenge for developing and transition country tax systems in attempting to produce sufficient revenues for needed services.

## 3.3 Tax Systems in Developing Countries: Current Issues

The tax systems of developing countries include the same basic tax categories used in developed countries: direct taxes on income and wealth; indirect taxes on consumption; property taxes; and trade taxes. The most common direct taxes are personal income tax, corporate income tax and wealth or inheritance taxes. The most common indirect taxes are value-added tax (VAT), and selected sales and excise taxes. Property taxes may be imposed on real property, such as land and housing, or on personal property, such as cars and boats. Trade taxes may take the form of import or export duties. For each type of tax, gender bias may exist explicitly in the tax laws and/or implicitly through the differential impact of the tax on women and men (Stotsky, 1997a). These biases will be discussed further in Section 5.

Table 3.1 describes the principal components of total government revenue: tax revenue, non-tax revenue and capital revenue.<sup>10</sup>

### **Table 3.1: Components of Government Revenue**

#### **Tax Revenue**

Income taxes (individual and corporate)
Payroll/social security taxes
Taxes on goods and services (VAT, sales, excise)
Property taxes
Trade taxes (import duties, export duties)
Other taxes

#### Non-tax Revenue

Income from public enterprises and property Administrative fees and charges Interest receipts Other non-tax revenue

#### **Capital Revenue**

Sale of fixed capital assets

Source: IMF, 2002

The distribution of tax authority between national, provincial and local government differs in each country. There are at least two aspects of how taxation is distributed across the levels. The first is who has authority to impose and collect it. The second is to which levels the money collected is distributed. Usually, broad-based taxes such as income tax and VAT are assigned to the central government, while geographically-specific taxes such as the property tax are local revenue tools. By spreading tax revenues across different tax instruments, ideally the fiscal system can better withstand economic fluctuations and can minimise the tax burden on any particular group of taxpayers or sectors of the economy.

A frequently used measure of the effectiveness of a country's tax system, and/or its tax competitiveness relative to other countries, is the ratio of total tax revenue to GDP. This ratio varies widely among both developed and developing countries. Sweden, for example, had a tax/GDP ratio of 51 per cent in 2002, whereas Australia and the US had ratios of 30 and 29 per cent, respectively, in 2001.<sup>11</sup> Among developing countries, there is also a wide range: 7.8 per cent in Bangladesh; 8.3 per cent in India; 10 per cent in Nigeria; 23.2 per cent in South Africa; 26.6 per cent in Jamaica; 31 per cent in Barbados; and 32.2 per cent in Botswana.<sup>12</sup> A low tax/GDP ratio may reflect an inadequate tax system and/or weak tax administration, or there may be other substantial non-tax sources of income, such as petroleum in Nigeria. Or it may be the result of conscious policy, as in South Africa where a national tax/GDP target was set in 1996 at no more than 25 per cent.

The following box provides a summary of the tax system in one country, Argentina, and its tax/GDP ratio.

## Box 3.1: Tax System Example - Argentina

The Argentinian tax system includes an income tax and various social security contributions, a value-added tax and a series of excise taxes, and trade taxes. These resources average 17–18 per cent of GDP. Provincial governments collect an additional 3 per cent of GDP in their own sales tax (a gross receipts tax (GRT)) and property and stamp taxes. There is a high degree of expenditure decentralisation, financed through a complex system of revenue sharing and transfers.

Source: Cuevas, 1990

Khattry and Rao (2002) study the use of different kinds of taxes according to the level of development. From a sample of 205 countries, they consider three main tax categories – income taxes, domestic taxes on goods and services and trade taxes – according to their relative shares of total tax revenue and of GDP at four levels of development ranked from low to high income. Their results are shown in Table 3.2.

As would be expected, poor countries derive a proportionally lower share of tax revenue from income taxes than high-income countries do, and they rely significantly more on trade taxes to fund essential public spending. For low-income countries, from 1970 to 1998 the share of trade taxes to total revenue fell from 40 per cent to 35 per cent, and the share of consumption taxes increased from 26 per cent to 33 per cent. The share of income taxes in low-income countries remained approximately the same over the period studied, constituting 27 per cent of total revenue in 1995–98.

Table 3.2 also shows the difference in shares of tax revenue at different levels of development. In 1995–98, trade taxes were 35 per cent of total revenue in low-income countries, 20 per cent in lower-middle-income countries, and less than 1 per cent in high-income countries. Domestic taxes on goods and services were about one-third of total tax revenue at all levels of development, with the share slightly higher in the middle-income countries.

Data for specific countries suggest that each country's distribution of tax revenues across types of taxes depends not only on its level of development, but also on its particular economic structure, the relative availability of alternative revenue sources and its unique social and political factors. Table 3.3 shows the relative shares of total revenue derived from income taxes, taxes on goods and services, property taxes and trade taxes for 22 selected Commonwealth developing countries. The income tax (personal and corporate) share of total revenue, for example, varies widely: from a low of 14–15 per cent in Bangladesh, Mauritius and Sri Lanka to a high of 48–60 per cent of total revenues in Malaysia, South Africa and Zimbabwe.

Table 3.2: Structure of Taxation and Level of Development

Income Group	Time Period	Income Tax (% of GDP)	Income Tax (% of Tax Revenue)	Domestic Taxes on Goods and Services (% of GDP)	Domestic Taxes on Goods and Services (% of Tax Revenue)	Trade Taxes (% of GDP)	Trade Taxes (% of Tax Revenue)	Indirect Taxes (% of GDP)	Indirect Taxes (% of Tax) Revenue
Low-income									
	1970–74	3.47	28.58	3.18	26.19	4.88	40.20	8.06	66.39
	1975–79	3.53	27.24	3.82	29.48	4.90	37.81	8.72	67.28
	1980–84	4.16	28.65	4.50	30.99	5.02	34.57	9.52	65.56
	1985–89	3.60	25.94	4.59	33.07	4.98	35.88	9.57	68.95
	1990–94	3.53	25.52	2.00	36.15	4.86	35.14	9.86	71.29
	1995–98	3.83	27.16	4.66	33.05	4.90	34.75	9.56	67.80
Lower middle-	4,								
	1970–74	3.57	24.37	4.06	29.09	5.05	34.99	9.11	64.08
	1975–79	4.64	27.32	4.15	26.06	6.05	34.70	10.20	92:09
	1980–84	5.20	29.82	4.55	27.80	5.27	30.03	9.82	57.83
	1985–89	5.27	29.64	5.28	31.76	4.59	26.33	9.87	58.09
	1990–94	5.27	29.79	5.77	34.07	4.81	25.16	10.58	59.23
	1995–98	4.79	27.30	7.40	41.85	3.80	20.18	11.20	62.03
Upper middle- income	ψ								
	1970–74	5.57	32.06	4.27	25.81	4.67	26.39	8.94	52.20
	1975–79	90.9	31.26	4.73	24.63	4.30	21.96	9.03	46.59
	1980–84	6.16	29.46	5.91	32.39	4.11	19.97	10.02	52.36
	1985–89	6.41	30.43	5.90	31.69	3.73	18.25	9.63	49.94
	1990–94	6.18	30.19	5.84	31.26	3.56	16.92	9.40	48.18
	1995–98	5.98	31.10	7.56	39.37	2.29	13.04	9.85	52.41

Table 3.2 (continued)

Group	Time Period	Income Tax (% of GDP)	Income Tax (% of Tax Revenue)	Domestic Taxes on Goods and Services (% of GDP)	Domestic Taxes on Goods and Services (% of Tax Revenue)	Trade Taxes (% of GDP)	Trade Taxes (% of Tax Revenue)	Indirect Taxes (% of GDP)	Indirect Taxes (% of Tax) Revenue
High-income									
	1970–74	8.81	35.94	7.73	31.54	1.66	6.77	9.39	38.31
	1975–79	9.52	35.97	7.94	30.00	1.32	4.99	9.26	34.98
	1980–84	10.36	36.41	8.67	30.47	1.07	3.76	9.74	34.24
	1985–89	10.74	36.54	69.6	32.97	0.75	2.55	10.44	35.52
	1990–94	10.37	35.40	9.62	32.84	0.46	1.57	10.08	34.41
	1995–98	10.80	35.36	10.20	33.40	0.25	0.87	10.45	34 22

Source: Khattry and Rao, 2002

Income group definitions: Low-income = GNP per capita of less than \$775, N = 62; Lower middle-income = \$756–2995, N = 55; Upper middle-income = Note: Income tax includes taxes on income, profits and capital gains; indirect taxes are the sum of domestic taxes on goods and services and trade taxes. \$2996-9265, N=38; High-income = \$9266 or above, N=50.

Table 3.3: Tax Structure of 22 Selected Commonwealth Countries

- `	Bangladesh 1999/2000	Barbados 2000/2001	Barbados Botswana 2000/2001 1997/1998	Ghana 1998	Grenada 1995	India 2000/2001	India Jamaica Kenya Malawi 2000/2001 1999/2000 2001/2002 2000/2001	Kenya 2001/2002	Malawi 2000/2001	Malaysia 1999	Mauritius 2001
Total revenue* Tax revenue*	203.70	1,703.00	8,169.00	3,551.60	205.00	2,039.00	90,373.00	196,613.00	2,039.00 90,373.00 196,613.00 20,880.00 1.886.00 75,965.00 160,394.00 19,285.00	59,157.00 25,374.00 47.917.00 21.642.00	25,374.00
Tax revenue	77.76	94.72	82.84	76.82	84.00	92.50	84.06	81.58	92.36		
(% of total revenue)											
PIT + Corporate (%)**	14.71	36.67		22.94	25.01	35.79	38.69	34.83		52.51	14.04
PIT (%) **				11.24		16.86			19.79		
Corporate (%)**				11.70		18.93			16.28		
Property (%)**		5.81			2.65					0.61	6.12
International trade	26.26	8.39	17.53	31.84	19.96	25.19	30.90	13.46	12.37	11.28	29.34
taxes (%)**											
Goods and	51.52	46.50	4.85	26.21	49.52	36.32	30.42	51.72	42.36	30.43	43.77
services (%)**											
Sales (%)**			4.85	20.38							
Excise taxes (%)**				5.83		36.32					

Source: Authors' calculations from IMF, 2002

Notes: The sub-components of tax revenue are not exhaustive; some taxes have been excluded such as levies, taxes on minerals and petroleum. Thus the columns do not sum to 100 per cent. Total revenue includes tax revenue and non-tax revenue but excludes grants.

kwacha; Malaysia: millions of ringgit; Mauritius; millions of rupees; Nigeria: millions of naira; Pakistan: millions of rupees; Papua New Guinea: millions of kina; Singapore: millions of dollars; South Africa: billions of rand; Sri Lanka: millions of rupees; Tanzania: billions of shillings; Trinidad and Tobago: \*Total revenue and tax revenue are in local currency. Bangladesh: billions of taka; Barbados: millions of dollars; Botswana: millions of pula; Ghana: oillions of cedis; Grenada: millions of dollars; India: billions of rupees; Jamaica: millions of dollars; Kenya: millions of shillings; Malawi: millions of millions of dollars; Uganda: billions of shillings; Zambia: millions of Kwacha; Zimbabwe: millions of dollars.

\*\*Stated as percentage of tax revenue.

Table 3.3: Tax Structure of 22 Selected Commonwealth Countries (continued)

	Nigeria	Pakistan	Papua New Singapore South Africa Sri Lanka Tanzania Guinea	Singapore	South Africa	Sri Lanka	Tanzania & Tobago	Trinidad	Uganda	Zambia	Zimbabwe
	2001	2001	2002	2001	1998/1999	2000	2001/2002	1999/2000	2001/2002 1999/2000 1997/1998 1998	1998	1997
Total revenue* Tax revenue*	2,227,236.00 470,402.00	) 535,091.00 3,222.00 ) 422,781.00 2,370.00	3,222.00	42,661.00 23,466.00	42,661.00 184.00 211,282.00 23,466.00 180.20 182,392.00	211,282.00	1,042.90	12,144.00 7,706.00	1,254.00	42,661.00 184.00 211,282.00 1,042.90 12,144.00 1,254.00 1,131,405.00 30,669.50 23,466.00 180.20 182,392.00 938.50 7,706.00 1,156.00 1,093,819.00 26,913.90	30,669.50 26,913.90
Tax revenue				55.01	97.93	86.33	89.99	63.46	92.19	89.96	87.75
(% of total revenue) PIT + Corporate (%)**	6)** 32.71	1 28.96		52.71	60.16	15.05	24.34	46.85	24.57	34.83	48.24
PIT (%) **										26.60	
Corporate (%)**										8.23	
Property (%)**		0.50		6.47		4.48		0.83			2.42
International trade	e 41.40	15.39		2.60	0.22	13.14	9.47	10.15	10.12	26.04	22.85
taxes (%)**											
Goods and	25.89	9 47.93		30.74	36.57	67.33	56.46	41.29	65.23	37.57	26.49
services (%)**											
Excise taxes (%)**	*						18.92		31.23	19.29	
<i>VAT</i> (%)**							37.54		34.00	18.28	

Source: Authors' calculations from IMF, 2002

Notes: The sub-components of tax revenue are not exhaustive; some taxes have been excluded such as levies, taxes on minerals and petroleum. Thus the kwacha; Malaysia: millions of ringgit; Mauritius: millions of rupees; Nigeria: millions of naira; Pakistan: millions of rupees; Papua New Guinea: millions of kina; Singapore: millions of dollars; South Africa: billions of rand; Sri Lanka: millions of rupees; Tanzania: billions of shillings; Trinidad and Tobago: \*Total revenue and tax revenue are in local currency. Bangladesh: billions of taka; Barbados: millions of dollars; Botswana: millions of pula; Ghana: billions of cedis; Grenada: millions of dollars; India: billions of rupees; Jamaica: millions of dollars; Kenya: millions of shillings; Malawi: millions of columns do not sum to 100 per cent. Total revenue includes tax revenue and non-tax revenue but excludes grants. millions of dollars; Uganda: billions of shillings; Zambia: millions of Kwacha; Zimbabwe: millions of dollars. \*\*Stated as percentage of tax revenue.

## 3.4 Tax Terminology and Concepts

#### 3.4.1 Tax Base

A country's overall economic base, to which its various taxes can be applied, is its land, labour, capital, mineral resources, and level of production and consumption. Some countries have unique resources such as oil and/or diversified economies, while others have a very narrow tax base, with only minimal economic resources or activities on which to draw for own-source revenues. For any specific tax, the tax base is the resources to which the tax rate is applied. To obtain a desired amount of revenue, the rate can be lower if the tax base is broad than if the tax base is narrow. A tax on a base of economic activities or assets that is more common to one sex than the other may result in gender bias.

#### 3.4.2 Tax Administration

The tax system must be consistent with each country's level of administrative capacity. Developing countries, especially, suffer from inadequate tax administration resources, weak public sector infrastructure, the lack of both quantity and quality of civil service workers, low public sector salaries and high levels of corruption. Thus, tax administration considerations are an especially critical issue for the design of tax systems in developing countries. There may be gender bias in tax administration if women, in their taxable activities, are more vulnerable to sexual harassment, bribes or other behaviours.

#### 3.4.3 Tax Burden

The tax burden is defined as the ratio of the tax payment to disposable income. Tax burden analysis is a critical tool for tax policy in order to evaluate the fairness, as well as the social and economic impact, of taxation alternatives. The tax burden can be calculated from data from tax returns by various categories such as income class, sector of the economy, individual vs. business, etc. Since the sex of the filer is not captured on tax forms, <sup>13</sup> gender tax burden analysis can be done by using assumptions based on demographic censuses and household surveys, or by matching sample tax information to other data captured by sex, such as social security information

#### 3.4.4 Tax Incidence

Tax burden measures tax payments according to who is remitting the tax payment by law. This is defined as the statutory incidence of the tax. However, the statutory tax-payer can, depending on market conditions, sometimes recover her/his tax payment by 'passing on' the cost of the tax to others. In the case of a business taxpayer, the business may recover taxes paid by passing on the cost of the tax in lower wages to workers or by charging consumers higher prices for its products. In this way, the true incidence of the

tax may fall on others. True tax incidence is difficult to calculate, so estimates are generally not part of formal tax analysis. However, consideration of tax incidence is an important consideration.

#### 3.4.5 Tax Incentives/Tax Expenditures

Many developing countries which are seeking to encourage business development and capital investment encounter pressure to provide tax incentives or exemptions, especially when they are competing for foreign direct investment. Evidence from years of local and national development efforts in both developed and developing countries show limited, if any, gain from such incentives. Socio-economic and political factors such as basic infrastructure, stable government, sound fiscal condition, available labour force and low social conflict are generally more decisive in influencing business investment decisions. In practice, however, decisions around tax incentives are often driven by political pressure.

#### 3.4.6 Tax Reform

Both developing and developed countries have engaged in periodic tax reform efforts over the last several decades. In developing and transition economies, tax reform has often been driven by international agencies such as the World Bank and the IMF, seeking to address countries' budget deficits and to open markets to globalisation. Their recommendations have resulted in the following reforms in most countries:

- Simplification eliminating minor taxes and consolidating others so as to reduce the number and complexity of taxes;
- Base-broadening bringing various forms of in-kind income into the base of the income tax and reducing special credits and exemptions;
- Rate reduction and harmonisation reducing top marginal tax rates and making these consistent across personal and corporate income taxes, and reducing the number of applicable tax rates and/or tax brackets;
- The creation of a VAT, often in a single-stage form, in order to provide a broad-based consumption tax with relatively simple administration;
- The reduction or elimination of import duties and export tariffs.

In the US, Western Europe and Australia, the emphasis of tax reform has been on reducing tax rates, and especially on providing tax relief to the rich and to businesses on the argument that this will stimulate investment and production ('supply side economics'). Developed countries rely heavily on income taxes, so tax reform has emphasised simplifying personal income tax rate structures, lowering top marginal rates, and bringing personal and corporate top marginal rates in line.

In both developing and developed countries there is concern that tax reform has adversely affected the poor both on the tax and the expenditure side. In developing countries, the increased reliance on indirect taxation, such as VAT, has raised concerns about regressivity. In developed countries there is evidence of an increase in the tax burden of the lower and lower-middle income groups and a reduction in the tax burden of the highest income groups. In both developing and developed countries there is also concern that there has been an increase in the relative tax shares paid by individuals through the personal income tax compared to those paid by businesses through corporate income taxes. Finally, in both developing and developed countries reductions in overall tax revenues have resulted in a 'fiscal squeeze' which can mean the reduction of needed public services with adverse effects in the short term on the poor and low-income groups and in the long term on overall social and economic development.

Tax reform has also posed two specific issues with explicit gender implications. First, the emphasis on tax simplification in both income and consumption taxes has led to policy recommendations to limit deductions and exemptions. Such limits have equity implications from both a class and a gender perspective. In the personal income tax, the restriction of certain deductions and exemptions which provide tax relief to women – for example childcare deductions, exemptions for dependents or deductions for insurance and pension contributions – may create gender inequity. In consumption taxes, the elimination of exemptions on products which are primarily consumed by women or are of primary importance to women could create gender bias. And in both types of taxes, base broadening which imposes a higher burden on the poor will also create higher burdens for women who are disproportionately poor in developed countries, and primarily poor in developing countries.

#### 3.4.7 Fiscal Decentralisation

An emphasis on fiscal decentralisation has been part of tax policy reform in developing and former socialist countries in the last few decades. The theory of fiscal federalism allocates expenditure responsibilities and tax authority between the various levels of government – national, provincial/state and local/municipal. The national government has expenditure responsibility for services that cannot be provided locally, such as defence; basic public institutions, infrastructure and services that address redistributional goals, such as equalising a basic level of development across regions; and 'merit goods' which the society deems should be provided to all citizens. Lower levels of government have responsibility for locally-provided services. The theory of fiscal decentralisation argues that citizens in a particular region or location can decide their own specific preferences for government services as indicated by their willingness to pay for those services through local taxes and fees.

Decentralisation can have serious equity implications if expenditure responsibilities

are devolved to lower levels of government which lack revenue instruments to support these responsibilities and if no money from national government or elsewhere follows. In many local areas of developing countries there is practically no tax base at all. In others, taxation of local bases such as property or agriculture has no relation to the taxpayer's ability to pay, creating an unfair tax burden on the poor and especially on poor women. In addition, a range of charges can occur at local level, which are often inequitable between and within localities.

In most systems of intergovernmental fiscal relations, there is sharing across levels of government for some services, such as health and education, but other services are a distinctly local responsibility, for example sanitation, and parks and playgrounds. Where there is sharing of expenditure responsibility with the national government, mid-level and local level governments receive grants or transfers from tax revenues collected nationally, and/or they may be authorised to apply a separate local tax rate to a tax base also used by the national government. Sub-national levels of government are also expected to rely on own-source revenues. Property taxes and user fees are the most common local source of revenue. But to cover costs, user fees are often set so high that they create a burden on the poor or limit access to basic necessities such as water.

The following section discusses key tax concepts that are used in tax analysis as a basis for the review of tax and gender in Section 5.

# 4

# Tax Policy Analysis: Concepts and Critiques

This section reviews key concepts that guide tax policy analysis. As stated previously, tax policy is at the heart of the political debate as to the level of public services that should be provided and who should pay for them. To engage in this debate from the perspective of gender equality in taxation, it is important to understand the concepts that are used, as well their limitations.

Tax policy analysis seeks to identify the impact of tax policy options on individuals, households, businesses, and economic growth and development, so that government decision-makers and the public can make informed policy decisions. Gender analysis of tax policy seeks to identify the differential impact on women and men of tax alternatives, to the extent that their social and economic roles and responsibilities are different, in order to ensure gender equity. In developing countries, most people – and especially most women – are poor, so the analysis of both income and gender equity is central to tax policy analysis.

## 4.1 The Behavioural Response to Taxation

The most basic concept in tax policy on which there is general agreement is that there is a behavioural response by individuals and businesses to changes in income and changes in relative prices. Taxation, with very few exceptions, alters both disposable income and the relative prices of both inputs to production and consumer goods, and thus affects a wide range of socio-economic decisions. <sup>14</sup> For example, decisions by men and women about the time they spend in formal, informal and unpaid work are influenced by the impact of taxation on wages and disposable income. Decisions about savings, consumption and investment are also affected by taxes.

The direction of the behavioural change depends on many factors and is open to debate, but there is agreement that behavioural responses must be carefully evaluated if tax policy is to attain the desired goal – whether in expected revenue or in the desired impact on targeted groups or activities<sup>15</sup> – with minimal unintended consequences.

Unintended consequences of tax policy are difficult to anticipate but must be considered. An example of unintended consequences from a revenue-raising measure used in India – although it was not a tax – is shown in the following box.

In developed countries with sophisticated data systems, revenue and impact analysis is performed by governments and academics according to various socio-economic and demographic characteristics, such as income class, ethnic group, geographical area, household type (married or single) and business sector/size. Attention to tax impacts by gender has begun only recently.

# Box 4.1: Example of Unintended Consequences – Alcohol Licence Fees as a Source of Revenue in India

India inherited from the British a revenue mechanism entailing the auction of licences to sell alcohol. In the 1980s, in the state of Andhra Pradesh, a populist regime came to power with promises of '2 rupees a kg. of rice' to be funded by revenue from the auction of alcohol licences. This coincided with a shift in the type of alcohol sold from higher end and more expensive liquor to a cheaper drink, arrak, also called 'the drink of the poor'.

To increase revenue from alcohol licences, the state government actively promoted the extension of licensing and the distribution of arrak throughout the state, and revenues from liquor licences increased by over 400 per cent in five years. The pervasive increased consumption of arrak, especially among poor men, increased rates of alcoholism and had a negative impact on household budgets and on family and social life in communities, including a rise in domestic violence.

In 1992, a group of women in Andhra Pradesh joined together to demand a ban on arrak sales in their village. The anti-arrak campaign soon spread to other districts and across the entire state, and the state government was forced to concede to the demand for prohibition. However, due to the need for revenues, and the absence of other tax mechanisms, state governments eventually reintroduced the alcohol licences.

Note: The Andhra Pradesh anti-arrak campaign is documented in the film When Women Unite: The Story of an Uprising by Nata Duvvury and Shabnam Virmani.

The following sections discuss three public finance concepts used in tax policy analysis: equity, efficiency and ease of administration. The first two, as discussed below, depend on underlying assumptions and normative values and there is no unanimity regarding their practical application, but they are important terms in tax analysis.

## 4.2 Equity

Equity in taxation expresses the idea that taxes should be 'fair', and is a concept used in all tax policy analysis. However, it should be noted that equity/fairness is a normative, value-based, concept and its interpretation differs across individuals, countries, cultures and time. Since it depends on one's particular perspective, as well as the specific circumstances being considered, the concept is difficult to apply in practice.

Tax equity is commonly discussed according to four definitions of 'fairness'. These definitions are also normative, and sometimes conflict, so they too are difficult to apply in practice. However, they are a common reference point for discussion.

Horizontal equity posits that taxpayers who are equally economically situated should be treated equally for tax purposes. Vertical equity posits that taxpayers who are not identical from an economic standpoint, but are differently situated, should be treated differently for tax purposes.

The limits in application of these principles are seen in the following examples:

**Horizontal Equity:** Two households both earning 60,000 rupees per month would be considered to be the same for tax purposes. But if one household earns this income through the labour of one wage earner and the other from two wage earners, are they 'the same'?

**Vertical Equity:** Of two individuals with an income of 50,000 pesos, one saves part of her income for retirement, while the other spends all of her income on consumption. The former will pay more taxes as she earns interest on her savings, while the latter may consume more government services since she is without private retirement savings. Is this 'fair'?

One measure for evaluating equity or fairness is the *ability-to-pay* principle, whereby those with a higher income should bear a larger share of the tax burden than those with a lower income. An alternative measure commonly used for user charges and local taxation is that of *benefits-received*, according to which it is fair to assess taxpayers in proportion to the benefits they receive from public services. From this perspective, those receiving the same benefits should pay the same, and those receiving higher or lower levels of benefits should pay more or less.

The limits of usefulness of these two principles in tax policy practice are seen in the following question: What is a 'fair' tax policy if a poor person, or an entire poor population, needs a great deal of public benefits yet has no ability to pay? The most commonly accepted idea of fairness in taxation is that taxes should be *progressive* – those with lower incomes should bear a lower share of the tax burden than those with higher incomes. But again the problem is in the application: what is the measure of who has a 'lower' income, and relative to whom? And how much less should they pay?

Progressive taxes are designed so that those with a lower income pay a lower percentage of their income in taxes than those with a higher income. Taxes that take a greater proportion of income from the poor than from the rich are said to be *regressive*. Income taxes can be made progressive by a structure of increasing marginal tax rates applied to higher brackets of income and/or through allowable credits and deductions and no-tax thresholds, which reduce the tax burdens of the poor. Consumption taxes are generally regressive, since the poor spend more of their income on consumption than the rich. Consumption taxes can be made less regressive through targeted exemptions, or lower rates for goods purchased primarily by the poor, and/or through special taxes or higher rates on luxury consumption items purchased mostly by the rich.

Gender equity in tax policy can be examined from several perspectives. First, according to the principles of horizontal and vertical equity, to the extent that women as a group or on average are situated similarly to men in terms of economic roles, behaviour or income, they should be treated similarly by the tax system; to the extent that they are situated differently, they should be treated differently. It is important to recognise that since gender interacts with race, ethnicity and geography, the concept of horizontal gender equity should be further extended along these lines. Secondly, since the vast majority of women in the developing world are poor, tax policies that address vertical equity and ability to pay will also improve tax equity for most women.

## 4.3 Efficiency

Taxes cost individuals and businesses through the loss of income which is transferred to government. If the income is 'recouped' by the same individual/business through public services, there is no net cost to the individual.

However, neo-classical public finance theory holds that, with the exception of lump-sum taxes, taxation imposes an 'efficiency cost' on society because individuals and businesses change what would otherwise have been 'optimal decisions' about labour, investment and production, and this deviation from optimality results in reduced overall economic output and growth. According to this theory, when taxes reduce social welfare – whether directly or indirectly – by more than the amount of revenues they produce, they are considered to be 'inefficient'. The loss to society constitutes a 'deadweight loss', also referred to as the 'excess burden' of taxation. Thus, good tax policy should produce the desired revenue and/or social goals of redistribution, environmental protection, etc., while minimising distortions to the economic decisions of individuals and businesses, and therefore minimising the cost to society.

It goes without saying that everyone wants the maximum social welfare possible – but again, the problem is in the application. Who defines social welfare? What is included in this concept? How is it measured?

The neo-classical calculation of efficiency assumes that there is a social welfare function that can be used as a benchmark for optimal output and production, against which the degree of inefficiency caused by taxation can be measured. Both mainstream and feminist economists have critiqued various aspects of this social welfare function:

- The underlying assumption that individual utility can be measured, compared and summed for society;
- The assumption that non-market time is leisure;
- The omission of non-market and reproductive activities with social value, in which women are heavily engaged;

- The omission of economic and productive value from public services;
- The omission of the possibility that taxation, through its redistributive mechanisms or through the public services it supports, can improve social and gender equity.

Feminist economists have especially criticised the neo-classical notion of efficiency. Diane Elson (1999), for instance, argues that efficiency is too often conceptualised and measured in ways that focus only on market-oriented production and ignore unpaid processes of social reproduction. This notion of efficiency can lead to policy actions which simply shift costs from the paid (public and private) economy to the unpaid economy of the household and community, with no genuine increase in efficiency in the sense of attaining maximum social welfare.

Taxation may also cause unrecognised economic inefficiencies if differential gender impacts are not included in the analysis. To the extent that women's economic roles place them in different positions than men, the impact of taxation on relative prices will be different for women than for men, and their decisions regarding labour supply, consumption, production and investment will be differently impacted. Specific examples of this are discussed in Section 5.

#### 4.4 Ease of Tax Administration

The third 'E' of tax policy, in addition to equity and efficiency, is that taxation should be easy to administer. Administration of a tax system must be funded from public revenue, reducing the amount of revenue available for other public services. In developed countries, the cost of collecting taxes has been estimated at 1 per cent of tax revenues, and in developing counties at possibly twice this. There may also be compliance costs to taxpayers in time and effort. To reduce the overall cost, the structure of the tax system should take into account the conditions of the country and its ability to administer and enforce the tax code. For many developing countries with conditions of low literacy, poor infrastructure and a weak civil service, consideration of the ease of administration is a particularly important factor in the design of tax policy.

## 4.5 The So-called 'Efficiency/Equity Trade-off'

Traditional public finance theory suggests that practitioners and policymakers should design taxes to raise needed revenues while addressing equity concerns and minimising economic inefficiency. Since equity and efficiency are in themselves normative principles, and often require a normative decision to determine the primacy of one over the other, the 'trade-off' between the two is a political decision to be decided by public debate.

This so-called trade-off between equity and efficiency is at the heart of many fiscal policy discussions. First, there is disagreement about the definitions of equity and

efficiency. Second, there is disagreement about the size or very existence of the tradeoff. In the economic growth literature, an emerging body of work suggests that inequality in the distribution of income and assets constrains economic growth and that a wider array of efficiency-enhancing redistributions exist in the policymakers' tool-box than have been recognised (Bardhan, Bowles and Gintis, 1998; Bénabou, 1996). Third, what is the relative value of a decrease in equity compared to a decrease in efficiency, and can this be determined?

Public finance theory may suggest that tax policy analysis should be guided by basic principles of equity and efficiency, but each country must determine through its political processes and within its own social and economic context how it defines these terms and their relative priority. And regardless of the answers, each country faces difficult tax policy decisions in seeking to raise revenues to support public services. The government can reduce the amount of targeted revenue in order to keep the tax burden low, but this in turn reduces the level and quality of public services that can be provided. An alternative is to collect the needed revenues by imposing a higher burden on wealthier taxpayers and/or on businesses, but this risks political opposition or the loss of the high-income tax base due to relocation, especially in conditions of liberalised open markets. A third alternative is to increase tax rates on all taxpayers, including those with low income, but this leads to an increased burden on the poor.

The role of good tax policy is to make sure that political decision-makers and the public have full information about these crucial decisions, so that they can be made in a democratic and transparent manner.

# <u>5</u>

# Gender Impacts of Taxation: A Review of the Literature

For this paper, we reviewed more than 67 studies that explored various dimensions of gender and taxation. These studies are summarised in Table 5.1. Most studies focus on a single country; far fewer provide comparative analyses of the gender dimensions of tax policy. The majority of the studies examine the gender dimensions of taxation in the industrialised countries, although there is a growing body of work on the gender dimensions of tax policy in countries like South Africa, Vietnam and Chile.

In the industrialised country literature, the topics covered most frequently are the impact of personal income taxes on labour supply, gender biases in systems of joint versus individual taxation and gender biases in social security taxes. The developing country studies focus on gender biases of indirect taxes and the gender dimensions of user fees.

## 5.1 Summary of Possible Gender Bias by Tax: Explicit and Implicit Bias

Because all tax systems evolve over time and reflect prevailing social and cultural norms which often discriminate against women, tax systems also contain gender bias. Stotsky (1997a) presents a useful conceptual framework for understanding this bias – both in its explicit and implicit forms. Table 5.2 shows the possible types of gender bias for four principal types of taxes: personal income tax; corporate income tax; commodity taxes; and trade taxes.

Explicit bias is found in specific provisions of the law that treat men and women differently. Explicit biases are relatively easy to identify since they depend largely on the language used in the tax code or tax regulations. They are more common in personal income tax arrangements than in other forms of taxation in both developed and developing countries. Implicit gender bias – different impacts on men and women because of different social arrangements and economic behaviour – may be found in personal income tax systems if they have joint filing requirements that tax secondary-earner incomes (primarily women's) at a higher marginal tax rate, thus affecting women's labour supply and other decisions. Implicit bias may also be found in other taxes such as consumption taxes, trade taxes and corporate taxes to the extent that they impact on female and male taxpayers differently, and in the way that payments are linked to the receipts of benefits under social insurance programmes.

Table 5.1: Summary of Literature Review of Effect of Taxation on Labour Supply, Household Production and Marriage/Fertility

Region	General		PIT		Corporate	Trade	Commodity	Trade Commodity Social Security User Fees Total	<b>User Fees</b>	Total
	Taxes	Labour Supply	Household Marriage, Production/ divorce, Time Use fertility	Marriage, divorce, fertility						
Global	_	_						_		m
0ECD	3	7	9	8			3	7		34
Developing	3									8
Sub-Saharan Africa	4	_	2			<u></u>		_	3	12
LAC	2					<del>-</del>	_	2		9
ECA	<del>-</del>	2			_			2		9
EAP						_	_	_		3
Total	4	11	8	8	_	3	ī	14	3	29

Table 5.2: Types of Gender Bias by Tax: Explicit and Implicit Bias

	Explicit	Implicit	Indirect/ Ambiguous Effect
1. Personal Income Tax			
A. Separate Filing			
i. Allocation of non-labour and/or			
family business income	Χ	X	
ii. Allocation of tax preferences	X	X	
iii. Rate structure	X		
B. Joint Filing			
i. Allocation of tax preferences	X		
ii. Responsibility for filing	X		
iii. 'Marriage tax' (marginal rates)		X	
2. Corporate Income Tax			X
3. Commodity Taxes			Х
A. VAT or broad-based sales		X	
B. Excise or selective sales		X	
4. Trade Taxes			Х
A. Import duties		X	
B. Export taxes		X	

Source: Stotsky, 1997

## 5.2 A Hypothetical Gender-Tax Typology

Table 5.3 is an example of a possible approach to gender analysis of taxation. The table presents a hypothetical gender-tax typology. The typology classifies countries according to their level of income: low, low-middle, upper-middle, and high income. The table profiles the primary economic activities of women in these countries and matches these to the principal types of taxes that affect women engaged in these activities. For each tax type, the most important potential bias, e.g. implicit or explicit, is described. Finally, the table includes a column with hypothetical priority recommendations for improving the gender equity of tax policy. Note, however, that the information in Table 5.3 is hypothetical and not derived from information on actual countries at these levels

The remainder of this section explores the various gender biases that may exist in different types of taxes. It summarises the findings of studies exploring the effects of personal income tax on female employment, household production and time use, and fertility in developed countries and draws lessons for developing countries. Although the literature from developing countries is not as extensive as the studies based on developed country data, the latter do provide background for the gender analysis of taxation in developing countries. The literature from developed countries can also provide important lessons about how gender bias can be identified and corrected in current tax systems, and possibly avoided in future tax policy decisions.

Table 5.3: Hypothetical Gender-Tax Typology

Level of Development	Principal Women's Economic Activities	Principal Taxes Affecting Women	Principal Type of Gender Bias	Principal Tax Recommendations
Very low	<ul> <li>Agriculture</li> <li>Market selling</li> <li>Small trade</li> <li>Home-based production</li> <li>Unpaid domestic work</li> </ul>	<ul><li>– VAT</li><li>– Selective sales taxes</li><li>– Property taxes</li><li>– User Fees</li></ul>	Implicit	<ul> <li>Simplify</li> <li>Low-income relief/credits</li> <li>No user fees for basic services</li> <li>Exemptions for food and basic necessities</li> </ul>
Medium low	<ul><li>Agriculture</li><li>Market selling</li><li>Small trade</li><li>Small services</li><li>Unpaid domestic work</li></ul>	<ul><li>VAT</li><li>Selective sales taxes</li><li>PIT&amp; Corporate taxes</li><li>Property taxes</li><li>User Fees</li></ul>	Implicit	<ul><li>Simplify income taxes</li><li>Low-income relief</li><li>Low tax rates</li><li>No user fees for basic services</li></ul>
Medium high	<ul><li>Small trade</li><li>Small services</li><li>Public sector</li><li>Factory work</li><li>Unpaid domestic work</li></ul>	<ul><li>VAT</li><li>Selective sales taxes</li><li>PIT &amp; Corporate taxes</li><li>Property taxes</li><li>User fees</li></ul>	Explicit	<ul> <li>Remove explicit bias</li> <li>Low-income relief</li> <li>Increase marginal rates</li> <li>Some targeted tax incentives/ credits</li> </ul>
High	– Full economic integration	<ul><li>VAT/Sales</li><li>Selective sales taxes</li><li>PIT &amp; Corporate taxes</li><li>Wealth &amp; Inheritance taxes</li></ul>	Explicit s s s s s s s s s s s s s s s s s s s	<ul> <li>Remove explicit bias</li> <li>Low-income relief</li> <li>Increase marginal income tax rates</li> <li>Some targeted tax incentives/credits</li> </ul>
	– Property taxes – User fees			

The sections below are organised by principal taxes. We begin with the personal income tax. The personal income tax constitutes a smaller share of total tax revenue in low-income countries than in developed countries but Table 3.3 shows that this share is as high as 20 per cent in Malawi, for example, and the share is likely to grow with development (Table 3.2). And finally, it is the tax whose structure can most easily address (gender) equity goals, so it is especially important for a gender analysis of taxation.

## 5.3 Personal Income Taxes

In developed countries, the personal income tax (PIT) raises desired revenues from a broad tax base with reasonably low rates. Its base of wage and non-wage income grows with the economy and keeps up with inflation. In countries with high literacy rates, the PIT is also fairly easy to administer. It can be used to address equity and redistribution goals through a progressive tax rate structure, with marginal rates which impose a lighter burden (or guarantee a minimum income) at the lower end of the income scale, and a gradually greater one at higher ends, following principals of vertical equity and ability to pay.

In developing countries, however, the PIT is often a more limited tax mechanism for several reasons: a) high rates of poverty mean that the personal income tax base is weak and narrow, and there is a greater need for redistribution; b) steep progressivity in marginal tax rates may unfairly burden those employed in the formal sector and/or create incentives for tax evasion and corruption by a small wealthy elite; c) high rates of illiteracy and a weak civil service infrastructure make the tax difficult to administer and enforce.

### 5.3.1 Gender Bias in Personal Income Taxes

The review of the literature suggests that most personal income tax systems are not gender neutral. Even in countries which have reformed their PITs to be more gender equitable, a number of thorny issues remain.

The appropriate unit for personal income taxes has been a subject of some debate in the economics literature (see Nelson, 1996 for a summary). The tax unit is the unit over which the income tax base is aggregated. In principle, the definition of the unit can vary for different types of income, although in practice the assessment unit is usually either the individual or a married couple. Some developing countries have a tax unit based on the extended family that is the typical social unit. In some countries with large Hindu populations, as in India, the 'Hindu undivided family' is the filing unit (Stotsky, 1997a). The Hindu undivided family consists of all male Hindus descended in the male line from a common ancestor, their wives and unmarried daughters; responsibility for filing for the family is vested in the eldest male member. Several countries have provisions in their tax code for multiple wives, and Nigeria's

tax code provides more generous tax treatment for the first wife than for succeeding wives.

Julie Nelson (1996) questions the definition of the 'proper' tax-paying unit and finds that both household and individual definitions are problematic. She argues that 'the notion of the household as a unit presupposes that households can be uniquely identified', which implies, for example, that information on different types of household must be available. Moreover, she suggests that equity principles (discussed above) dictate that levels of household wellbeing must also be well-defined; as noted earlier, this is difficult to do in practice. The notion of the individual as the unit, on the other hand, tends to focus attention on adult earners as the only tax-relevant human beings, ignoring issues of dependency and social relationships. Nelson looks for a middle ground, suggesting that the tax code should define a unit of taxation as an individual earner plus his or her dependents, i.e. those persons who are unable to support themselves for reasons such as youth, old age or chronic disability.

Although discussion of the details of Nelson's proposal is beyond the scope of this paper, two observations are worth noting: her proposal is marriage neutral, and it imposes a lighter tax burden on the two-earner couple than the one-earner household. If countries are to adopt this approach, many questions, such as the treatment of property income, how to allocate child exemptions and the allocation of social security taxes and benefits will need to be worked out in the tax code. One possible line of approach is the calculation of an 'adult consumption equivalent', which includes both working adults and their dependents.

There are two types of personal income taxes – schedular and global. A schedular tax calculates liability on each source of income, while global taxes apply a single rate schedule to aggregated income. Schedular taxes are typically found in developing countries, especially those where tax administration capacities are weak. Global income taxes are typically found in industrialised countries and increasingly in developing countries. Schedular regimes may contain implicit gender bias, insofar as they include elements, such as deductions and credits, based on the personal characteristics of the taxpayer (Stotsky, 1997a).

Global income taxes, which are assessed on either an individual or a family basis, contain a number of explicit gender biases. Under an individual filing system, married individuals file a separate return based on their own labour earnings. The allocation of non-labour earnings and exemptions or deductions for children and other purposes is determined by the country's tax laws and can be a source of gender bias. Most developing countries require individual filing for at least some sources of income. Joint filing systems with progressive marginal rate schedules also contain explicit gender bias in the allocation of tax preferences (i.e. to only one spouse) or in filing rules. Most industrialised countries had joint filing at one time, although many have moved to individual filing in order to reduce gender discrimination.

## 5.3.2 Gender Implications of PIT Individual Filing

Stotsky (1997a) identifies three types of potential gender bias in systems that permit individual filing: the allocation of non-labour or business income; the allocation of tax preferences or exemptions; and tax rates.

In income tax systems with individual filing, wage income is attributed to the worker, but the attribution of non-wage income in a household is not so simple. Tax codes allocate this income in a variety of ways, including attributing it to the higher earner, allocating the income equally between partners, allowing couples to make the decision for how to allocate the income or attributing the income to the spouse who possesses legal ownership of the property. Many countries' tax codes allocate all non-wage income to the husband whether or not this is true in practice: this is the case for some countries in Latin America, Asia and Africa which allow women to pay tax on their labour income but attribute all non-labour income to the husband.

In many parts of the world, family business income is also attributed to the husband regardless of the spouse's role in the business: this is the case in Tanzania, for instance. Often the rationale is to prevent tax avoidance. However, there are administrative solutions to this problem that would be more gender equitable.

A second form of gender bias occurs in the allocation of exemptions. Often, countries give exemptions or deductions for a variety of purposes, including for dependent children and non-working spouses. These exemptions must be allocated between spouses or parents in some way, and rules differ widely across countries. In Jordan, for instance, if husbands and wives file separately, some deductions are only available to the husband and not to his wife. In Zimbabwe, married men with non-working wives are entitled to a special credit but there is no such provision for married women with non-working husbands.

A third form of gender bias can be found in a country's tax rate structure. Countries may levy different tax rates on men and women, with a higher rate being applied to married women than to married men, as was done in South Africa until 1995 and still happens, for example, in several other countries in the Middle East. This will be discussed further below.

# 5.3.3 Gender Implications of PIT Joint Filing

In the US and in most countries in Europe which tax the household, the income obtained by all family members is added together, resulting in a higher marginal rate than the one which would be paid if an individual filed alone (de Villota and Ferrari, 2001). The literature review showed that higher marginal rates deter women from entering the paid labour market and keep them in part-time jobs. According to Sainsbury (1996), joint taxation stems from an ideology favouring the traditional family where the husband is identified as head of the family and financial supporter of the household, and the wife devotes her time exclusively to looking after children and

domestic work. High marginal rates still exist in many PIT systems despite more gender-equitable mechanisms such as income splitting, the family ratio, a joint rate with lower tax levels for families than for single persons without dependents or deductions for accrued tax obligations.

Beyond high marginal tax rates for secondary earners, Stotsky (1997b) identifies two other forms of explicit gender bias in systems with joint filing. One form is making tax preferences available only to the husband, for instance, an allowance for married men if he supports the household but not an allowance for married women. Many countries in the Middle East and North Africa allow men to claim deductions for children on their income. These deductions may only be taken on women's income if she is the sole breadwinner, if her husband agrees (Jordan), or if the father is dead or handicapped (Lebanon). The rationale behind these rules is that a family should only be allowed to take the deduction once, even if both the father and mother work. A second form of bias occurs in countries that require jointly filed returns to be submitted in the name of the husband; consequently wives have no separate existence as taxpayers. This was the case in Britain until 1990; Switzerland continues the practice today.

# 5.3.4 Impact of PIT on Married Women's Labour Supply 17

As noted above, systems of joint filing with a progressive marginal rate schedule often discourage secondary earners because the tax on the secondary income starts at the highest marginal rate of the 'primary' earner. This is known as the marriage penalty.

The public finance literature on the marriage penalty in the US and in European countries concludes that the principal impact of high marginal rates has been on wives, who generally earn less than their husbands and are in effect taxed at their husband's marginal rate (Bartlett, 1998). Thus, wives generally receive less after-tax income on each dollar they earn than their husbands, which is a significant disincentive to married women to enter the paid labour force. The disincentive effects of high marginal tax rates on married women's labour supply are further exacerbated by their responsibilities in the unpaid care economy during their reproductive years. <sup>18</sup>

The empirical literature also shows that the labour supply elasticity of married women, who are often assumed to be secondary earners, is greater than the labour supply response of married men. The 1981 US Tax Act reduced the marriage penalty by instituting a secondary-earner deduction. Feenberg and Rosen (1995) found that following the Act married women's participation in the labour market expanded by almost enough to pay for the revenue loss caused by the deduction. Eissa's (1995b) analysis of the 1986 US Tax Reform Act, which lowered the top marginal tax rate from 50 per cent to 28 per cent, similarly found that married women responded more strongly to the increased work incentive than men did. A Norwegian study found an increase in the labour supply of married women with equalised tax schedules across

marital status (Aaberge *et al.*, 1989). The implication of this research is that to minimise the 'deadweight loss' of the income tax, everything else being equal, married women should be taxed at a lower rate than other workers.

However, it is important to note that the conventional literature examining the impact of taxation on labour supply assumes that the alternative to paid work is leisure, whereas most women allocate their time to home production and to leisure. The findings of this literature are most certainly incorrect: when household production is incorporated into the analysis, the labour supply response of married women is seriously underestimated (Apps, 2002b). In addition, purchased childcare is considered to be a cost only when the mother seeks paid employment: the time cost of childcare provided by mothers is assumed to be zero. Analysing data from European countries, Apps (2002a) finds a strong positive effect of childcare subsidies offered through the tax system on female labour supply and on fertility because market childcare and mother-provided care are close substitutes. Further research is needed to show the relevance of this effect for women in different income classes in countries at different levels of economic development.

Finally, tax and welfare policies that ignore household production can lead to an increase in inequality across and within households and impose significant efficiency losses by reinforcing gender roles (Apps and Rees, 1999a). We turn to this issue next.

## 5.3.5 Impact of PIT on Time Use and Household Production

Models of the effect of taxes and transfers must rely implicitly or explicitly on some theory of how households function. The traditional view, pioneered by Gary Becker in 1974 and 1981, argues that men specialise in market work and women specialise in household production. Other things being equal, labour in household production is subject to diminishing returns – if the value of household capital and technology remains constant, the value of goods and services produced by an additional unit of labour begins to decline at some point. Both men and women making decisions about time allocation devote their labour to household production as long as the value of hourly product from that production is greater than the market wage. Women therefore work longer hours in the household than men and fewer hours in the market. Many studies of the impact of tax and transfer in developed countries rely on this theory of the household.

Many scholars have criticised this view and its use in tax analysis. Apps (2002b) and Apps and Rees (1999a, b and c), for instance, point out that specialisation in domestic production may result either from higher productivity or from lower productivity in household work. As Apps (2002b) says, 'High productivity in domestic production implies a low implicit price and a higher demand for domestic output. The higher the productivity, the less time is required to produce a given domestic output and more time is available for market labour supply. The net effect depends on the

price elasticity of demand for domestic output in relation to domestic productivity.' Thus, tax policy analysis that is based on average household income may be misleading because it misrepresents the true distribution of living standards.<sup>19</sup>

As noted earlier, one measure of tax fairness is ability to pay. Julie Nelson (1996) makes the point that 'if household services are counted as income, the ability to pay of an earner and homemaker couple with a given money income will be higher than that of a household in which earning the same money income requires that both adults work outside of the home'. (This is similar to the example we discussed in Section 4.) The two-earner couple has less time for leisure and needs more money to pay for market goods to replace those formerly provided at home. Nelson argues that households with an earner and homemaker 'should also be compared, in terms of household services, to the household of an earner and a child or other dependent (based on age, disability or illness) who have need for care'. She notes that the ultimate losers in systems where domestic output is 'free' are lone parents, unless they receive special help in other ways.

Because the norm of marriage and the nuclear family as well as the allocation of women's time between market work, unpaid labour and leisure in developing countries differs significantly from that of women in developed countries, the relevance of the European and US literature is questionable. More research is necessary to understand the impact of tax and welfare systems on women's time allocation and unpaid work in developing countries. Nonetheless, one lesson from the developing country literature stands out. Basing the design of a personal tax and transfer system on combined family income, while ignoring women's unpaid work in the care economy, is problematic.<sup>20</sup> This type of system can widen gender inequality and increase the gap between rich and poor.

## 5.3.6 Impact of PIT on Fertility

Concern about population decline has induced some governments to adopt explicitly pro-natalist policies: these countries have included Canada, Singapore, France, East Germany and Hungary. At the other extreme, China has instituted economic incentives to reduce fertility. Given that most developing countries are trying to reduce fertility, the effects of tax incentives on birth rates in these countries remains unclear and there is currently no literature on this issue.

A study from the US by Whittington *et al.* (1990) examined the impact of the personal exemption in the US tax code on fertility. Using fertility data from 1913 to 1984 to ascertain whether changes in the exemption level impacted on fertility rates, they found that the personal exemption had a positive and significant effect on the national birth rate. They note that increases in the personal exemption served as effective decreases in the costs of raising children. Although the elasticity of the birth rate with respect to the exemption was not large, this study suggests that countries can influence to a degree the fertility decisions of its citizens through changes in tax

policies. But putting a tax on additional children would be unfair to the poor, who tend to have more children.

## 5.4 Payroll Taxes

Payroll taxes are an alternative form of income tax. They are deducted at source from the wages of employees, often with a matching portion paid by the employer, based on total payroll. Developed countries use payroll taxes primarily to fund social security systems. Many developing countries use payroll taxes as a source of general revenue because they are relatively easy to administer and collect, as well as to fund social security systems. However, a payroll tax generally imposes a greater tax burden on the poor than the PIT because: a) it falls on a much narrower base than the PIT – only on the wage portion of income and only on the wages of those employed in businesses over a certain size which are required to keep formal accounting books; b) because of the narrower base, it requires much higher tax rates than the PIT in order to raise the same amount of revenue; c) the employer may be able to recoup all or part of taxes paid by reducing wages to workers or increasing prices to consumers, causing an additional burden on the poor; and (d) it falls on all employees, whereas low-income workers in these companies may be exempt from the personal income tax.

Generally, people pay taxes on their income when they actually earn it. However, an exception is made with respect to taxation of income contributed to retirement funds and pensions: tax is paid when the retirement or pension benefits are actually received after the worker retires. The intention of the tax deferral is to encourage individuals to save for retirement. However, its result is regressive because the deferral is only available to people with sufficient income to be in the income tax system and because, within that group, those who earn more and are taxed at higher marginal rates benefit more from deferral than those who earn less and pay lower marginal rates.

In the past ten years, countries with defined 'pay-as-you-go' (PAYG) public pension systems have been moving to privatised defined contribution systems. Pension reforms are well underway in Latin America and the transition countries in Eastern Europe. In these countries, a multi-pillar structure is developing. The new structure consists of two mandatory pillars. One pillar handles the funded accounts that people are required by government to have. This pillar is usually defined contribution instead of defined benefit. The benefits are funded rather than 'pay-as-you-go' and the funds are privately managed. The second pillar is a publicly managed tax-financed arrangement designed to provide a social safety net to protect low wage earners.<sup>21</sup>

There are several rationales for defined contribution plans. First, they are seen to reduce the incentive for evasion, which is a problem in many countries. James (1998) notes that, in some countries, between 20–30 per cent of the covered labour force evades taxes. She argues that smaller taxes and closer links between benefits and

contributions reduce the incentive for evasion. Second, defined contribution plans are structured to discourage early retirement, which gives the system more financial sustainability. Third, defined contribution plans avoid large payroll tax increases which may be necessary over time as a country's population ages.

Critics do not agree that privatising pensions is the best way to raise revenues and support workers in their old age. Baker and Kar (2002) contend that it is also possible to raise the return on defined contribution systems by adding more government funds which is, in effect, what private accounts do: they increase government payments to support pension systems, contributing general revenue (and imposing a tax on capital income) but they do it through changing relative prices. Apps and Rees (2002) and others also show that the switch from PAYG to funded pension systems in developed countries is not a solution to the problems raised by increasing aged-dependency ratios and declining fertility in those countries.

## 5.4.1 Gender Bias in Payroll Taxes

Our review of the literature found several studies on social security taxes in the developed countries.<sup>22</sup> A few recent studies have also explored the effect of pension reforms in Latin America and Eastern Europe on women and men.

Three main factors explain gender gaps in private pension coverage: women's household responsibilities, women's work patterns and pension design. Because pension entitlements are predominantly through work, women's responsibilities for unpaid care work, as well as their predominance in informal employment and seasonal and part-time jobs, all restrict their access to the private pension sector. Private pension coverage is more extensive in larger firms and in those industries requiring a skilled, stable and full-time labour force. Pension design is the third factor, and many pension systems contain both implicit and explicit gender bias.<sup>23</sup>

Several studies in our review examine the gender impact of pension reforms in Latin America. Baker and Kar (2002) note in their survey of Latin American pension reform that reform efforts have included raising the retirement age to the same level for both sexes. While this is a reasonable policy, it means that women workers will experience a larger reduction in benefits than men. The impact of this benefit reduction is compounded by the shift to defined contribution plans which the authors claim are less favourable to women for two reasons. First, since women generally earn lower wages than men and spend fewer years in formal employment, they are likely to accumulate much less in individual accounts than men. Second, the structure of the benefit payment hurts women. The benefit payment from a defined benefit plan is gender blind, with a worker's benefit depending on his or her contributions. However, the payments from most defined contribution plans are based on gender-specific life expectancies. This means that a woman who retires at a particular age receives a smaller benefit than a man retiring at the same age with an identical accumulation

because she has longer life expectancy. The combination of gender-specific annuities and the lower accumulations that women are likely to have could lead to reductions in benefits for many older women.

In contrast to these claims, James *et al.* (2002) undertook a cross-country empirical analysis of the gender impact of pension reforms in Chile, Argentina and Mexico. They found that women generally accumulate retirement funds and own-annuities that are only 30–40 per cent those of men from the defined contribution pillar of the multi-pillar systems. However, they note that the effect seems to have been mitigated by targeting the new public pillars toward lower wage earners, many of whom are women, and by restrictions on pay-out provisions, including joint annuity requirements, which redistribute income from husbands to wives within households. They claim that as a result of this 'total lifetime retirement benefits for women reach 60–80 per cent of those for men; for "full-career" women, they equal or exceed benefits of men'.

Cox Edwards (2001) provides a detailed analysis of pension reform in Chile. She finds that it 'increased women's incentives to participate in the labour market, to save, and to use the social security system as a channel for their savings ... Under the new system's rules, there is no minimum level of contributions to obtain a pension; contributions at a young age get an increased weight via compound interest; and widows can keep their own pension in addition to their widow's pension. These three characteristics raised the marginal benefit of own contributions for working women relative to what the old system offered.'<sup>24</sup>

Because the specifics of pension reform vary across countries and there are too few studies to draw clear conclusions, further research is necessary to determine whether women are better off in defined contribution plans. Public pension schemes which can provide insurance and redistribution on a wider scale than personal pensions may be an important supplement to defined contribution plans in countries undergoing reform (which are largely middle income), or be a better alternative for women in countries at low levels of income, where informal employment and widespread poverty co-exist.

# 5.5 Commodity Taxes (VAT, Sales, Excise)

Since the time of J.S. Mill and Alfred Marshall, economists have argued that taxation should be imposed on consumption, not income, in order to avoid the disincentives to paid labour, investment and savings which income taxes create. However, commodity taxes impose a greater tax burden on the poor than on the rich because the poor spend most or all of their income on basic consumption. Commodity taxes also alter the relative prices of taxed and untaxed goods and thus alter individual and household decisions about consumption, and business decisions about investment and produc-

tion. Commodity taxes generally seek to apply the lowest rate possible to the broadest possible tax base, with minimal exemptions.

The value-added tax is a commodity tax that has become the predominant form of commodity taxation in many developed and developing countries.<sup>25</sup> By April 2001, 123 countries had some form of VAT; 27 countries in sub-Saharan Africa had a VAT, compared to only four in 1989 (Ebrill *et al.*, 2001, Table 1.1).<sup>26</sup> The most populous countries without a central VAT are India and the US. Developing countries that have adopted a VAT have relatively higher GDP per capita and rely somewhat less on international trade than countries without a VAT.

The incidence of VAT and other broad-based consumption taxes falls in principle on the final consumer. Any preferential treatment, as a result, is usually given to the final consumer. Typically, preferential treatment is applied to goods and services that are considered necessities, such as food and medical care, so as to minimise the burden on the poor, and to goods and services that for administrative reasons are difficult to tax, such as financial services (see Box 5.2). Tax preferences can also apply to certain purchasers or producers, such as non-profit institutions or the government. Most countries with a VAT exempt agriculture, because a large share of the agricultural sector is informal and most of the poor are active in that sector (Le, 2003).

## **Box 5.2: Case Study on Trinidad and Tobago**

Trinidad and Tobago incorporated in its VAT regime numerous zero and reduced rates and exemptions that were intended to make the system less regressive than the previous purchase tax. Zero rates were granted to basic goods, in addition to exports, and exemptions included health-related services, most of education, rental of residential property, and bus and postal services. Even with these poverty-relief features, the VAT is seen as a successful revenue-raising tool.

Source: Le, 2003

There is some disagreement in the literature about whether and how to include the informal economy in the VAT base. Le (2003) observes that VAT tends to impose high compliance costs on small traders who generally do not have sufficient resources to keep proper records of their transactions and to comply with accounting rules. Although the number of small traders is large, including them in the tax net would produce limited revenue and create high administration costs.<sup>27</sup>

## 5.5.1 Gender Bias in Commodity Taxes

Commodity taxes such as VAT alter relative prices between taxed and untaxed goods. Gender biases in such taxes tend to be implicit rather than explicit. Elson (1999) and

others point out that gender biases can result from women's differential consumption patterns. Although the exact nature of these patterns must be discerned in a specific country context, generally it has been found that women tend to consume goods and services that benefit family health, education and nutrition, while men consume more of their income on personal items. Thus, women may bear a disproportionately larger burden of indirection taxation. At the same time, Ingrid Palmer (1995) argues that the exemption of a range of essentials can turn a VAT into a modestly progressive tax. Since men and women partly purchase and produce different goods and services, she suggests that VAT exemptions can be used as a policy instrument to advance gender equity.

Commodity taxes also alter relative prices between the paid and the unpaid care economy and in so doing affect the distribution of work between them. This distribution has clear gender implications.<sup>28</sup>

The literature review found two studies that examine the gendered effects of indirect taxation. Smith (2000) provided evidence showing the VAT burden on South African households at different levels of income who consume the same commodity basket. His analysis showed that expenditure on VAT as a proportion of both total taxes paid and annual income is highest for the lowest income household and becomes lower as household income rises. The South African VAT does give special consideration to the distributional impact of the tax. It zero rates certain foodstuffs, including brown bread, maize meal, dried beans, milk powder, rice, vegetables and fruit. Some goods that are important for the poor, such as paraffin, which is widely used by poor women for cooking, have been zero rated in recent years, and the South Africa's Women's Budget Initiative can claim that it contributed to this change. Poor women bear more of the burden of indirect taxes, both because they are poor and because of gender differentials in consumption patterns.

The second study examined the effect of indirect taxes on women-owned businesses. Van Staveren and Akram-Lodhi (2003) explore the effect of VAT in Vietnam on women's enterprises. Because female-owned enterprises have a different input cost structure than do male-owned enterprises, the VAT places a relatively higher burden on female-owned businesses than male-owned businesses because it increases their total costs and lowers their overall profits.

Female entrepreneurs have three types of input costs: labour, capital and materials. On the labour side, female entrepreneurs utilise less paid labour and less unpaid household labour than do male entrepreneurs as a result of gender stereotyping which structures the division of household responsibilities. Value-added tax is a proportion of the estimated value added that is generated by the enterprise. When male-operated household enterprises have more inputs of unpaid (female) household labour, there is an uncosted 'invisible' input into the production process that generates value added which will affect the calculation of VAT. Van Staveren and Akram-Lodhi conclude that gender inequality is not a function of the absolute amount of VAT that is payable,

but rather a function of the gender division of labour in which female-operated enterprises have less access to unpaid household labour and lower levels of value added and which therefore results in female entrepreneurs paying a relatively higher proportion of their total earnings as tax.

On the capital side, women entrepreneurs face higher capital costs than do male entrepreneurs, in part because of credit market imperfections in Vietnam, which reduces their earnings. The Vietnamese government calculates VAT payments on 'reasonable' costs of capital, which is likely to proxy the market interest rate for loans. Female entrepreneurs pay higher interest rates than the market rate, so their VAT payments are likely to be higher than those of male entrepreneurs.

On the material input side, Van Staveren and Akram-Lodhi note that industrial and occupational sex segregation influences the burden of VAT payments. Womenowned enterprises are more likely to be in the trade sector, while male entrepreneurs are more likely to be in the production sector. Female entrepreneurs are, therefore, more likely to face the 10 per cent VAT levy, while male entrepreneurs pay the reduced VAT rate of five per cent. Moreover, the authors note that the food and beverages sub-sector, where women predominate, is in the 20 per cent VAT band and this rate is applied to food stalls in markets and on the streets. Thus, profit rates for female entrepreneurs may be diminished by the higher rates of tax applied to women in this sector.

The Vietnamese VAT contains 27 categories of exemption, including one for businesses below a threshold defined as 'the minimum level of business turnover minus a reasonable cost of business operation'. Many female-operated businesses fall below this threshold, and in principle they are exempted from VAT. However, exemptions are only granted to businesses that are registered and the registration rate for business owners is lower for women than for men. Thus, Van Staveren and Akram-Lodhi conclude: 'Non-registration by female-owned SMEs will considerably reduce the net earnings of an enterprise because they will be paying tax on higher priced inputs for which no redemption of VAT payments on inputs will be available' (Van Staveren and Akram-Lodhi, 2003, p. 26).

# **5.6 Corporate Income Taxes**

Developed and developing countries generally impose an income tax on the income of business entities which are legally organised as corporations. There is a great deal of controversy in tax theory regarding corporate taxation since technically corporations are conduits for shareholders and corporate income is passed on to shareholders as dividends or as increased stock value. A corporate income tax means that the same income is taxed twice – both as personal income of the individual shareholder and as corporate income. Despite the 'double taxation' argument, corporate taxes continue

to be applied because they produce significant revenues for governments which would be difficult to replace without a shift in tax burden to some other sector. In many countries, reductions in corporate tax rates have resulted in a shift of the tax burden from businesses to individuals. In South Africa, for instance, Hartzenberg (1996) and Smith (2000) report that there has been a continuing shift from company to personal tax. A traditional concern with regard to the corporate tax, particularly if there are high corporate rates, is the possible negative effect on business investment, although there is debate in the literature on the magnitude of the effect.<sup>29</sup>

A particular issue for all countries in open, globalised economies, and especially for developing countries which desperately need tax revenues but want also to attract foreign investment, is taxation of multinational corporations. In tax policy, there is general agreement that corporations should not be taxed by multiple jurisdictions on the same income, but there is no agreement as to whether corporate income should be allocated for tax purposes according to 'source' or 'residence'. The former allocates income to where it is produced; the second allocates income to where the entity has its residence or domicile.

Grunberg (1998) notes that the question of how to tax foreign income poses a trade-off between equity and revenue objectives. Adopting a residence principle is best for raising revenue: foreign income is treated as domestic income and taken into account in calculating the applicable tax rate. In practice, foreign income is difficult to track because of difficulties in decoding foreign tax documents, lack of co-operation at the operational level, legal challenges and competitive behaviour among countries. Hence, the alternative, the source principle, is often adopted, mostly for corporate taxation. So Treasuries try to make up the shortfall caused by the difficulty of collecting corporate taxes by increasing taxes on immobile factors such as land and labour (e.g. through payroll taxes).

Another issue is that there are many ways that multinational corporations allocate revenues and costs (transfer pricing) in order to minimise their total tax payments to all jurisdictions; this can be particularly challenging for local and national governments. This is particularly true in countries where capital markets have been liberalised and corporate investment can easily be moved to other countries. In the corporate tax, policymakers face most clearly the dilemma of balancing the need for short-term revenue to fund services which are needed for social and economic development against the need for capital investment which is also needed for economic growth.

## 5.6.1 Gender Bias in Corporate Income Taxes

Since corporate taxes apply to legal entities, they tend not to contain explicit gender bias, but an implicit gender bias is possible, depending on the incidence and behavioural effects of corporate taxation. Preferential tax treatment of different industries may affect men and women's employment and/or consumption differently. For instance,

a differential tax on mining may affect men's employment more than women's as men predominate in that industry.

Gender issues may also be reflected in the distribution of the tax burden between corporate taxes and individual taxes, and in the overall level of taxes, if reductions in tax rates mean reductions in services on which women depend more than men. The literature on corporate taxation concludes that corporate taxes are borne by both the owners of capital and by consumers; there has been no attempt to tie these effects to gendered outcomes.

Only one study in this review discussed the gender dimensions of corporate income taxes. Smith (2000) notes that in principle women could benefit in two ways from increased investment that is associated with reductions in company taxes. They could presumably share in the profits resulting from these investments; however, in South Africa the majority of shareholders of large companies (which receive the greatest benefit from the reduction of corporate taxation) are men, so women may not be direct beneficiaries from reduced corporate taxes. Second, women could potentially benefit if corporate tax relief encourages private sector investment that creates employment. However, Smith observes that private sector investment in South Africa has been primarily in capital-intensive industries such as IT, energy and oil, in which employment is dominated by men.

In other countries, foreign direct investment (as a result of tax breaks for multinational corporations) has been associated with increased employment for women, especially in light manufacturing industries and in services. There is considerable debate, however, about the quality of these jobs and the long-term sustainability of this type of employment for women.

# 5.7 The Tobin Tax Proposal<sup>30</sup>

The Tobin tax, originally proposed by Nobel Laureate James Tobin, seeks to reduce volatility in the global economy by imposing a small international tax on foreign exchange transactions. There are now a variety of proposals for Tobin-type taxes, which many argue could be a significant source of revenue that could be used for development purposes (Palley, 2001; Haq *et al.*, 1996). According to some estimates, a Tobin tax has the potential to raise a large amount of revenue. For instance, using 1995 currency transactions figures, Felix and Sau (1996) estimate the global revenues from a Tobin tax of 0.1 per cent to be in the range of \$148–180 billion. If the tax were set at 0.05 per cent, the revenue estimate increases to \$90–97 billion.

Many consider revenues from Tobin taxes as an attractive policy alternative given the widely acknowledged problem of tax competition (Tanzi, 1996; OECD, 2001) which has contributed to an erosion of national tax bases and the shift from direct to indirect taxation (Rodrik, 1997). It is argued that a Tobin tax would be relatively pro-

gressive in incidence, because the burden would fall predominantly on higher income individuals and on corporations. The amount of revenue raised would also depend on the extent to which the tax reduces currency speculation. If the tax has little impact, the revenues will be relatively larger; if the tax has a large impact, the revenues will be relatively smaller. However, many argue the tax is justified (Palley, 2001; Baker, 2002; Pollin *et al.*, 1999).

The literature on Tobin taxes has discussed problems of enforcement, avoidance and evasion that plague national tax systems. Critics argue that these problems make the Tobin tax infeasible. Advocates respond that evasion and avoidance should not be the determining factor in determining the feasibility of such a tax, because every tax system is subject to some evasion and avoidance. The extent of such behaviour is always a concern. Other factors relevant to the implementation of a Tobin tax include the amount of needed revenue that the tax raises and the behaviour it discourages. Palley (1991) argues: 'This is the test that should be applied to the Tobin tax – just as it should for all tax systems – and on this test the Tobin tax scores well.'

Beyond these issues is a question about the nature of regulation in a dynamic global economy. Critics of the Tobin tax argue that financial markets will innovate to avoid it. Advocates respond that this may be true but, again, that this does not mean that a Tobin tax is unwarranted. There are many ways to regulate undesirable behaviour of firms.

### 5.7.1 Gender and Tobin Taxes

It has been suggested that a portion of the revenues raised by a Tobin tax could be awarded to governments for the design of gender-equitable social insurance and social protection systems. A Tobin tax would also potentially reduce market volatility – caused by movements in speculative, short-term capital flows – which has disproportionately negative impacts on women (see Cağatay, 2003).

## 5.8 Trade Taxes

Taxes on foreign trade take the form of import and/or export duties. They are similar to broad-based domestic consumption and excise taxes. They typically apply to a broader range of commodities than excise taxes but with different rates. Trade taxes (import tariffs and export taxes) are important policy instruments for resource allocation purposes (i.e. protection for import-competing sectors) and for revenues. Historically, developing countries have relied heavily on trade taxes because other tax bases and tax administration capability are weak, and trade taxes are relatively easy to assess and enforce. Though the relative weight of trade taxes has declined over the last two decades because of trade liberalisation policies, trade taxes continue to be an important source of revenue for the governments of developing countries, especially in low-income countries. The share of total trade taxes in total tax revenue between the

1970s and 1998 averaged 36 per cent for low-income countries, 29 per cent for lower middle-income countries, 19 per cent for upper middle-income countries, and only 3 per cent for high-income countries (see Table 3.2 above).

## 5.8.1 Gender Bias in Trade Taxes

There are very few studies of the impact of gender bias in trade taxes. One study, produced by the South Africa Women's Budget Initiative, examines the gender dimensions of customs and excise taxes. By analysing a range of regional and multilateral free trade agreements, Tanya Goldman (2000) explores the impact that South Africa's import and excise taxes are likely to have upon sectors with concentrations of women. The paper delineates the individual trade agreements and the potential repercussions that adoption could have upon women, particularly if the agreements are implemented suddenly. In recent years, South African trade policy has shifted towards export-led growth. The abrupt rescinding of earlier tariffs has forced many low-skilled industries (where women are disproportionately concentrated) to rapidly become much more competitive.

Goldman identifies three ways in which tariffs affect women: as workers in sectors of the economy where goods are imported and exported (e.g. clothing and agriculture); as consumers of imported goods (e.g. medicines); and as traders in export goods. In South Africa, women predominate in labour-intensive industries, such as clothing and textiles, which have been hard hit by import tariff reductions. The reduction of import tariffs on basic goods such as clothing and food, resulting in lower prices, would be a benefit for poor women. These benefits, however, must be weighed against job losses in the affected industries and the spillover effects to the overall economy. Stephanie Seguino has pointed out that tariff cuts that lead to job losses can lead to a fall in wages in non-protected sectors as workers from protected sectors seek employment. Further, depending on the multiplier effects, economic downturns can set in, causing widespread job losses and cuts in public expenditure.

Rao (1999) has examined trade taxes and human development in developing countries. Although he does not explicitly consider gender issues, one can draw at least one gender implication from his analysis. Rao demonstrates empirically that trade taxes (as a percentage of trade volume) have a positive effect on current government expenditure. Reductions in import taxes, on the other hand, were associated with decreases in public sector investment in infrastructure and education, which in turn could have negative implications for women's time allocations and expenditures.

## 5.9 Tax Reform and Gender

As noted in Section 3, tax reform can have gender implications. First, the emphasis on tax simplification in both income and consumption taxes leads to policy recommendations to limit deductions and exemptions. Such limits have equity implications from

both a class and a gender perspective. In personal income tax, the restriction of certain deductions and exemptions which provide tax relief to women, for example childcare deductions, dependent exemptions or deductions for insurance and pension contributions, may create gender inequity. In consumption taxes, the elimination of exemptions on products which are primarily consumed by women or are of primary importance to women could create gender bias. And in both types of taxes, base broadening which imposes a higher burden on the poor will also create higher burdens for women who are disproportionately poor in developed countries, and primarily poor in developing countries.

The literature on tax reform efforts bears out some of these conclusions and also notes other effects. Reform efforts in Japan's PIT in 1989 did indeed exacerbate gender inequality. Shibata (1994) showed that changes to the rate and deduction structure discouraged married women from entering paid employment. At the same time, some reform efforts in European countries, including France, the Netherlands and the UK, have sought to reduce explicit gender bias in personal income tax codes. Reforms in the US reduced the marriage penalty and eliminated other forms of gender bias in exemptions and deductions (Nelson, 1996). Some developing countries have also taken explicit measures to reduce gender biases in personal income taxes. Malaysia, for example, changed its tax system in 1991 from one in which the income of married women was attributed to their husbands unless she elected for separate assessment to a system in which husbands and wives are treated as separate taxable units. In 1995, South Africa unified rates schedules for married persons, single persons and married women, reducing the rates for the two latter categories.

Smith (2000) analysed the gender implications of South Africa's efforts to reform direct and indirect taxes. In 1994, the Katz Commission was appointed to review the tax system and make recommendations for improvement. The commission placed great emphasis on equity as a key principle, along with concerns about poverty and income inequality. Some of the changes that have taken place since 1994 eliminate formal discrimination based on gender, including the introduction of a unified income tax rate structure for individuals and the removal of the distinction between married and unmarried households; tax relief has also been given to low- and middle-income taxpayers through adjustments to tax rates and income brackets.

Smith points out, however, that these changes have not eliminated all forms of discrimination against women. The new system still discriminates against households with only one income earner which pay over four times as much income tax as households with two income earners and the same income and number of dependents as the one-earner household. Gender bias also continues to exist in the way that tax deductions are permitted in the new system. Allowances for travel and accommodation are biased toward men. Although reforms in the PIT closed some of the most egregious gaps between women and men, it did not close gaps between rich and poor individuals

and rich and poor households. Since women in South Africa predominate in informal employment or work in low-paid formal sector jobs, relatively few women have benefited from a more progressive income tax system. More recent reforms seeking to give relief to low-income individual earners may, however, correct these biases.

Smith also evaluated the impact of reforms in consumption taxes on women and the poor. He noted that VAT contains exemptions for certain foods and basic goods and services consumed by low-income people, but that the tax reform commission rejected further extensions of zero rating, fearing substantial revenue losses.

As Esim (2000) notes, the gender implications of tax reform require further research. While the links between gender equality and personal income taxes are better understood, the implications of indirect and corporate taxes still need to be investigated. More country case studies need to be analysed to give a better picture of the direction of the reforms and how they influence women.

## 5.10 User Charges

As noted in Section 3, as part of fiscal decentralisation, sub-national levels of government are increasingly expected to rely on own-source revenues. Property taxes and user fees are the most common local source of revenue. Since the review did not find any articles on the gender dimensions of property taxes, we restrict our discussion to user fees.

User fees are charges for services or products provided by governments. Following the recommendations of the World Bank and the International Monetary Fund in the past two decades, user financing of basic social services have become common practice in many developing countries, and user fees have become an alternative to tax-based financing for public services. A critical debate in developing countries centres on whether user charges should be imposed on basic public services like water, electricity, health and education. User fees that are more commonly accepted are for selected locally provided optional services such as public transportation charges and parks and recreation.

Proponents of user fees argue that governments need revenues and that charging users for services is an efficient way to raise money for these services. They also argue that user charges may be an effective method of reducing consumption of scarce resources. This argument is based on the idea that user fees allow governments to impose the cost of services on the citizens who use them, and allow market forces to set an economically efficient level of services. Additionally, proponents argue that user fees are especially appropriate when state or local governments provide services that also are provided by the private sector, particularly if they are not core government services.

Opponents of user fees argue that there is growing evidence of the equity losses: reduced utilisation of services and negative effects on wellbeing and health resulting

from the introduction of user fees. In most cases user fees have the unintended effect of decreasing access to basic services (education and health, in particular) by the poor. Opponents also claim that user fees for health and education do not appear to generate adequate revenue to support even basic provision of these services. National user fee systems have generated an average of only around 5 per cent of total recurrent health system expenditures, gross of administrative costs. The cost recovery levels for health services range from 0.5 per cent in Burkina Faso (1981) and Guinea Bissau (1988) to 9 per cent in Lesotho (1991–92) and Mozambique (1985) (Reddy and Vandemoortele, 1996).

Palmer (1995) notes that user charges can be highly regressive, as they constitute a tax on services which were formerly often free at the point of delivery. They have the effect of transferring the costs of providing these services to women in the unpaid care economy. User fees also frequently exacerbate discrimination against girls with respect to utilisation of healthcare and education. One way to mitigate this is to apply user charges to services where gender disparities would not be significantly worsened by attaching a price tag. Another way is through exemptions.

In some countries, governments allow exemptions for the poor from user fees. This can be done in two ways. One is through direct or individual targeting. This type of targeting identifies poor individuals and provides them with services for which others are expected to pay at reduced or no cost. The second is indirect or characteristic targeting. Under this approach, the focus is not on particular individuals but on general characteristics of the programme or its clients. An example of this is the exemption of particular services important to the poor, for instance, immunisation or maternal and child health programmes. The services in question are provided free of charge to rich as well as to poor people, but the poor are expected to benefit most because they are the primary service users and/or gain the most from the services.

Two criteria have been established to evaluate the effectiveness of a targeting programme. One concerns the percentage of the poor population reached by the programme: a programme that fails to cover a large percentage of the poor population is commonly said to suffer from under-coverage or from a large exclusion error. The second criterion is related to the percentage of the programme's benefits that go to people who are not poor. A programme that is designed to serve the poor, but that is used primarily by the non-poor, is considered to have a major leakage problem or large inclusion error.

Vandemoortele (2000) cites a study that found that exemptions based on the ability to pay are extremely uncommon in practice. Exemption schemes for health in sub-Saharan Africa are not only rare, but they are also implemented in informal and ad hoc ways. The decision to exempt is often left to the discretion of local service providers, while the characteristics of the poor are generally not clearly defined. Further, if financial incentives or staff performance appraisals are linked to the

criterion of cost recovery, as they have been in many countries, then there can be a direct trade-off between the goal of revenue collection and that of reducing the negative impact of user fees on the poor.

Nanda's (2002) comprehensive review of the literature on the gender dimensions of user fees in the health sector in Africa during the mid-1990s brings the above discussion into sharp relief. She notes that a gender-based analysis of user fees should look at two aspects: the differential impact of user fees on utilisation of healthcare between men and women; and how user fees affect women's utilisation of healthcare for themselves. The limited nature of the data makes it impossible to answer the first question, so her review focused on studies that examine the second issue. Nanda cites studies suggesting that the introduction of user fees is often followed by a subsequent fall in service utilisation. The magnitude of this drop is often greater and occurs over a longer period for the poor. In Kenya, the introduction of user fees (amounting to half a day's pay for a poor person) in government outpatient health facilities led to a dramatic reduction in utilisation of STD services by both men and women, but the fall was significantly greater for women. Fewer women than men utilised the services prior to the introduction of user fees. Nine months after their introduction, the fees were revoked and women's utilisation rose to a greater level than the pre-fee levels. Use of maternal healthcare services was also affected when fees were introduced or revoked. Although some basic antenatal care and family planning services are exempt from user fees in some countries, there was little incentive for providers to apply exemptions, and abuses and inconsistencies in application were widespread.

The World Bank has recently acknowledged that user fee exemption systems seldom work in practice. In a letter to a member of the US Congress dated 29 September 1999, World Bank Vice-President Eduardo Doryan said: 'Experience in and since the 1980s has shown that the poor have not been effectively protected in many cases [from user fees]. Planning for new or higher fees has frequently outstripped adequate protection and implementation of exemptions and safety nets.' As a result, the World Bank now accepts that user fees should not be imposed for basic services such as education and health (World Bank, 2003).

# Tax Policy Debates and Issues

The literature review in the previous section shows that there is a wide range of ways in which tax policy can have a gender-differentiated impact. This section discusses the principal tax policy debates of special relevance to developing countries, some general tax issues for developing countries and specific issues which are of importance to a gender analysis of taxation.

## **6.1 Tax Policy Debates**

Tax policy is at the heart of an ongoing fiscal debate in all countries, which is of special importance to developing ones, about the role of government and the appropriate size of its budget. This debate recurs in every budget cycle. Its resolution, each time, is fundamentally a political process, which should occur with full participation of all sectors of society, full information and complete transparency, but which is too often determined by politicians and elites behind closed doors. The fiscal debate is posed around three interrelated questions:

- 1. What should be the overall size of the government budget? This question requires an evaluation of available resources, the desired level of public expenditures and the acceptable level of possible deficits.
- 2. Of the needed resources, how much should be raised from taxation (government revenue) and how much should be raised from other sources, such as foreign aid (grants) or borrowing (financing)?
- 3. Of the required amount of tax revenues, what should be the distribution of the tax burden across sectors and taxpayers?

Tax policy is also at the heart of the social debate regarding how to achieve goals of income and gender equity. Two factors have driven a new interest in the use of tax policy to actively promote and achieve redistribution goals. First, there is concern about increasing income inequality in both developed and developing countries around the world. Secondly, there is growing frustration and scepticism about the ability of public expenditure to reach the very poor. Some feminist economists, particularly, have argued that tax policy should be considered as a tool for achieving gender equity.

# 6.2 General Issues of Tax Policy in Developing Countries

In all countries, and especially in the developing world, the primary issue with regard to taxation is the complex task of designing policies that will provide sufficient

revenues to support desired social and economic development needs, while avoiding policies which may discourage the investment in financial and human capital which is also fundamental for economic growth.

A second issue for developing countries is that a large proportion of the population is too poor to pay taxes, creating a very narrow effective tax base. The narrower the tax base, the higher the rate has to be, which can result in greater incentives for tax evasion and corruption – a vicious cycle for these countries. Additionally, income and consumption can only be taxed to the extent that they can be reached by the tax net, and many developing countries suffer from weak tax enforcement and administration. Tax officials in developing countries need sufficient resources to improve tax administration and enforcement, but allocation of these resources from own-source revenues is difficult – requiring either higher overall taxes, or the reduction of other expenditures.

A third set of issues concerns the need to harmonise tax, trade and industrial policies. Taxes and tax rates need to be harmonised with neighbouring states to avoid tax competition, and to take into consideration the impact of open trade markets. This is especially important for small developing countries. However, in order to reduce trade taxes, countries must create other revenue sources, and the shift to increased reliance on consumption taxes has not resolved the problem of budget deficits. Are there other, more gender-equitable, options that can produce desired revenue?

Finally, there is a link between taxation and other means of closing deficits. Tax alternatives must be considered in the full context of other financing alternatives, including the acceptable level of debt and the need for foreign aid. This paper has not explored that important issue.

# **6.3 Gender-specific Tax Policy Issues**

Beyond the general issues of tax policy for developing countries, there are a number of unresolved issues that are specific to the gender dimensions of taxation. The following questions are especially relevant to countries at low and low-middle levels of development:

- How should tax policy deal with rural—urban differences, low levels of formal employment and relatively large informal sectors, which is where the majority of women workers are concentrated?
- How can tax administration be made appropriate for low levels of literacy, especially
  for women, the impact of the HIV/AIDS epidemic on households and income, and
  weak tax enforcement and capacity?
- How should tax policy deal with different family structures, which influence the tax unit treatment (such as extended families, more than one wife and lone parent households)?

- How can tax benefits/allowances be designed to flow to the mother, since evidence suggests that a greater family benefit then ensues to the household?
- How can implicit bias in indirect taxes heavily used in developing countries be
  assessed, given the problems of measuring differential consumption? There is great
  difficulty in showing non-neutrality in value-added taxes and other broad-based
  consumption taxes.

Two other issues apply more generally to countries across levels of development:

- How should tax systems recognise the value of unpaid care-giving services? What
  types of exemptions should there be for spouses or partners who contribute unpaid
  care work?
- Should sex-based versus gender-neutral annuity/mortality tables be used for social security and other tax payments?

## 6.4 Summary of Debates and Issues

Tax policy analysis can play an important role in informing tax policy debates and addressing income and gender equity issues, but it is the overall impact of both taxes and expenditures in improving the distribution of income and wealth and improving gender equity that is important. The impact of both tax and expenditure policies on poverty, income distribution and gender equity require considerable more study in a wide range of countries in order to bring empirical evidence to bear on crucial policy decisions.

# Methodological/Analytical Tools for Gender-revenue Analysis

This section discusses possible methodological approaches and analytical tools for gender revenue analysis. The first part reviews the standard tax analysis methodologies used by tax practitioners to evaluate the impact of tax proposals on individuals and on specific sectors of the economy and their relevance for gender analysis: tax burden analysis; tax incidence analysis; representative taxpayer analysis; microsimulation models; and macrosimulation models. The second part discusses methodologies applied to expenditure analysis by gender responsive government budgeting (GRGB) initiatives and their applicability to gender revenue analysis.

## 7.1 Traditional Methodologies and Tools for Tax Policy Analysis

## 7.1.1 Tax Burden Analysis

Tax burden is defined as the statutory tax payment obligation as a percentage of disposable income. For a particular tax, or for a combination of taxes, to the extent that there are data on tax filers, tax payments and net income, tax burdens can be calculated by income class, by sector of the economy and by individual compared to business filers. Gender tax burden analysis would require data by sex. Tax filers, even when they file separately, are not identified by sex, but gender burden analysis could be done with appropriate assumptions based on census or other demographic data, which all countries have in rudimentary form.

# 7.1.2 Tax Incidence Analysis

Tax incidence analysis can be used to analyse the distributional impact of taxes and subsidies, for instance how this varies for various groups of households on the basis of income, geographic location and other factors. Gender incidence analysis takes into account intra-household relations. Gender incidence analysis, as with burden analysis, may require assumptions based on limited data, but can nevertheless be a useful exercise.

Incidence studies define the groups of interest, typically in terms of income/consumption, geographic location, gender, ethnicity, age and socio-economic group, and then calculate the taxes paid by each household group. <sup>32</sup> Incidence studies distinguish between statutory incidence (those who have to transfer the tax to the government) and economic incidence (those whose real purchase power declines because of the tax). To quantify the tax paid, the technique either estimates the taxes paid as the official tax rate multiplied by the pre-tax value of expenditure, or estimates the 'effective' tax rate by dividing the actual tax revenues by the tax base and applying this to the categories of interest.

# Box 7.1: The Feasibility of Gender Incidence Analysis: The Case of Mongolia

A recent report by Richard Marshall for UNDP reviewed the feasibility of conducting a gender incidence study of the tax system in Mongolia. Marshall's review found that suitable sex-disaggregated data is not available for an incidence study of direct taxes; the authorities simply do not hold sex-disaggregated data for any tax. Even where personal records are held, they are not sex-disaggregated. Investment in primary data collection is a necessary first step in the development of a gender revenue analysis in Mongolia.

Source: Marshall, 2003

## 7.1.3 Micro-simulation Modelling

Tax policy analysis often uses micro-simulation modelling, based on a representative sample of tax return data. Such modelling requires a sufficiently large base of micro data on individuals, such as from taxpayer records stripped of personal identifiers, or from current population surveys or household surveys. These models are particularly used for personal and corporate income taxes. By capturing key variables in the database which interact with tax policy options, and based on defined assumptions about taxpayer behaviour, the micro-simulation model can calculate the effect of tax policy options on the number of filers, size of payments, impact of special credits or deductions, distribution of burden and total revenue collected. Since tax data do not capture the sex of filers, micro-simulation modelling by gender would need to make assumptions based on other data about the distinct economic activity of women and men and intra-household allocation decisions. The reliability of the results would depend on the data used and the underlying assumptions required.

# 7.1.4 Macrosimulation Modelling

Another potential tool for understanding the gender dimensions of revenue is a macrosimulation model. Such models involve longer-term forecasting and are based on macro variables such as inflation rates, trade balances, forecasts of profits across different economic sectors, estimates of elasticities and so forth. Macrosimulation models aim to produce a gender-aware medium-term macroeconomic framework. Some common models include financial programming models, fixed coefficient, two-gap growth accounting models, macroeconometric simulation models and computable general equilibrium models. Most of these are currently gender blind, but a variety of approaches could be used to introduce gender concerns, including gender disaggregation of variables already included in the models, introduction of new variables and equations which capture the pattern of gender relations, and construction of models

which include a sector that represents the social reproduction system (see Çağatay et al. (1995) for a detailed discussion of gender-aware macroeconomic models).

Fontana (2002 and 2001) has developed gender-aware Computable General Equilibrium Models (CGE) for Bangladesh and for Zambia.<sup>33</sup> These models include a reproductive sector and leisure in addition to the market sector, and treat male and female labour as two distinct factors of production. While the Fontana analysis simulates changes in trade policy in an economy where male and female workers have different levels of consumption, it could be extended to simulate the effect of a variety of tax reform measures.

# 7.2 Methodologies and Tools Used in Gender-responsive Government Budgeting

Gender-responsive government budgeting has developed five tools for gender expenditure analysis, used in pre- and post-budget analysis (Budlender *et al.*, 1998). The first three are used in pre-budget analysis, and have been developed through participatory methods and surveys of opinions. If adapted for gender revenue analysis, these would be:

- 1. Gender-aware medium-term tax policy framework;
- 2. Gender-aware tax policy appraisal;
- 3. Gender-disaggregated tax burden assessment.

The other tools are used in post-budget expenditure analysis and are similar to the methodologies discussed previously for tax analysis. These require significant resources to implement successfully, as they do for expenditure analysis, but they would include:

- 4. Gender-specific tax incidence analysis;
- 5. Gender-aware revenue statement.

# 7.3 Summary

All methodologies for gender analysis of taxation require sex-disaggregated data. Three types of data are needed for this analysis: data on time use of individuals within households, which are important in estimating the productivity of domestic or household work; data on household budgets and the degree of sharing of household income and expenditure by gender; and sex-disaggregated data on both formal and informal employment.

The analysis of tax impacts is a complex task and would require significant investment of resources and time. A thorough analysis of the gender impact of taxation requires consideration of both direct and indirect effects. It should also be paired with a gender analysis of expenditures to properly identify the full gender-differentiated cost-benefit ratio of public policy.

The type and reliability of available data drive the methodology chosen in all tax policy analysis. If the aim of the analysis requires a level of accuracy which cannot be achieved with available data, interim strategies must be devised to produce the needed data. More descriptive methodologies can be used in the short term.

The most appropriate methodology for gender revenue analysis will depend on the following interacting considerations:

- The goal of the analysis and its expected policy impact;
- The type of gender bias, i.e. whether it is explicit or implicit;
- The data available;
- The degree of accuracy expected;
- The cost-benefit ratio of performing the analysis.

The first goal of gender revenue analysis should be to eliminate explicit bias by removing legal inequities in the tax system. The methodology appropriate for this goal is a detailed legal review of the tax laws to reveal bias and proposals for statutory changes.

The second goal should be to reduce or eliminate implicit bias in tax policy. Attaining this goal may require complex methodologies and significant investment, so the cost-benefit should be evaluated on a case-by-case basis. For example, implicit bias in personal income tax marginal tax rates (the 'marriage tax'), or in the allocation of deductions and credits between husband and wife, would ideally require simulation modelling. Analysis of the incidence by gender of consumption taxes, or of other taxes that are passed on throughout the economy, would require macrosimulation or general equilibrium modelling. Analysis of implicit gender bias could also benefit from new theoretical work on household bargaining models. Pilot studies in selected countries would be a very useful next step.

A third goal for gender revenue analysis might be to advance gender mainstreaming in the public sector. In this case, descriptive methodologies that are low-cost but serve to raise awareness may be the best approach. Gita Sen (2000) discusses a number of methodologies for gender mainstreaming in ministries of finance that could be adapted for this purpose.

Finally, there could be an attempt to improve public sector equity and efficiency by advancing understanding of the impacts of public policies on both women and men. This goal could be met by some of the participatory methodologies used for gender expenditure analysis, and revenue analysis could perhaps be integrated into existing procedures. Tax literacy can be beneficial for civil society and women's organisations, and gender sensitisation is especially important for tax planners and staff who work in ministries of finance in all countries.

# **Conclusions and Recommendations**

Tax systems are not gender neutral. They contain both explicit and implicit gender biases. Gender revenue analysis is an important tool for revealing and correcting this bias. Such analysis can advance the commitment made by governments in the 1995 Platform for Action following the UN World Conference on Women to incorporate a gender perspective in budgetary processes as a means of supporting gender equality and development programmes that enhance women's empowerment.

This paper provides a preliminary review of the ways in which tax systems may be gender biased. It argues that gender revenue analysis is a fruitful tool for both analysing and correcting gender biases. It concludes with some specific recommendations through which gender revenue analysis could be advanced:

- 1. Support and expand existing efforts to advance and improve the collection of sex-disaggregated data in countries around the world. This should include support for tax policy units in ministries of finance to collect, process and analyse sex-disaggregated as well as other needed data, and produce tax analysis reports. Tax administrators might also consider collecting information on filers by sex. Gender budget analysts and activists could create a list of standard data that should be collected in order to carry out adequate gender analysis in their particular country.
- 2. Support a legal review of tax law in developing countries to identify explicit bias and formulate recommendations for change. Within countries, or at a more aggregate level, gender budget initiatives should consider developing a list of areas of the tax code with potential gender bias to be examined and corrected.
- 3. Support research on the equity and gender improvements that could be garnered if a greater share of tax revenue is shifted from indirect to direct taxation which can be made more progressive. These studies should focus on the distributional consequences, as well as on administrative and implementation aspects, of personal and corporate income taxes. For those countries where personal income taxes represent a larger share of total tax revenue, gender budget advocates should look at potential biases in individual and joint filing, as well as biases in the structure of exemptions, deductions and allowances.
- 4. Support research on gender bias in indirect taxes such as VAT, consumption taxes and trade taxes. Gender budget analysts should examine the nature of exemptions provided under VAT and whether these exclude small taxpayers, and food and basic necessities that contribute to human capital development, including health and education.

- 5. Introduce pilot gender revenue initiatives in a small number of developing countries where minimum data requirements can be met. These gender revenue initiatives could take the form of a gender revenue analysis report, following the model of tax expenditure reports which were adopted by many taxing jurisdictions during the 1980s. In these reports, the limits of data and assumptions for each type of analysis are clearly explained, but the issues are identified and quantified to the extent possible in order to inform public decision-making.
- 6. Support further research on a typology that would match a country's level of development with the principal taxes it uses, and the principal possible gender biases inherent in these taxes, with recommendations on how to address these biases and make tax systems more gender equitable. Such a typology could be developed and expanded by policymakers and practitioners in gender budget training exercises.

This paper has focused on taxation and user fees. Further work could usefully explore the impact of local taxes, such as property taxes and licence fees, on gender equity. Further work is also needed on the gender impacts of foreign assistance, since many very poor countries depend for a large share of their revenue on foreign assistance. Finally, another important focus for research could be gender analysis of government financing options, including domestic and foreign borrowing.

#### **Notes**

- 1 At present more than 50 countries have some form of gender budget initiative (Budlender and Hewitt, 2002).
- 2 Gender is conceptualised as the social meanings given to biological sex differences.
- 3 Folbre (1995) profiles the change over the eighteenth century as housework disappeared from official view in the national accounts of the US and European countries.
- 4 Data for 1995 to 1999 reveals that in New Zealand women with no children spend 34 hours per week on unpaid work. This figure increases to 50 hours per week for women with one or more children. Men's unpaid labour remains constant at 23 hours per week, regardless of the presence of children in the household (United Nations, 2000).
- 5 See Sen (1990) and Agarwal (1995).
- 6 Braunstein and Folbre (2001) remark that between the seventeenth and nineteenth centuries, husbands in Great Britain and the US enjoyed explicit control rights over their wife's property and person, and over the products of her labour.
- 7 Notable exceptions are Australia, South Africa, Uganda, Tanzania and the United Kingdom.
- 8 Distinct from revenue are two other categories in government finance statistics: grants, which include foreign aid; and financing, which includes domestic and foreign loans and printing money.
- 9 For a description of one of the most egregious cases, see King Leopold's Ghost by Adam Hochschild.
- 10 See definition of revenue, as distinct from grants and financing, in note 8.
- 11 OECD, 2002.
- 12 Years vary; see IMF, 2002.
- 13 Name, and possibly other information such as a tax identification number or social security number, is required on a tax form, but sex is not required.
- 14 The only taxes which do not distort economic decisions are those which can be imposed 'lump-sum', where no behavioural action can be taken which will avoid the tax. The standard example of a lump-sum tax is a head tax imposed per person. In Britain, Margaret Thatcher's government attempted to impose a head tax, which drew widespread popular protest and was withdrawn.
- 15 Targeted tax goals may be to benefit a particular group, for example those at the lower end of the income spectrum or those with children; or they may be intended to penalise socially undesirable behaviour and reward the opposite, such as environmental taxes or benefits.
- 16 Specifying household wellbeing requires the aggregation of individual preferences into a household utility function, a procedure which has been criticized (see Nelson, 1996; Folbre, 1997 and 1994).
- 17 We found no studies on the impact of PIT on marriage and divorce in developing countries. There is a large literature in the US on this topic. The US tax system imposes a marriage penalty for a two-earner couple when they marry if their earnings are close to even, but if one spouse earns very little and the other a lot, the tax burden is not very different than that for the single-earner couple. This balance has changed over time with various reforms to the income tax. Eissa and Hoynes (1998) find that recent reforms in the US tax structure, as well as an increase in secondary earners, have dramatically changed the tax and transfer consequences of marriage; however, this impact is heterogeneous across income brackets. Other research found a correlation between the marriage penalty and the number of cohabiting (as opposed to married) couples (Bartlett, 1998). When the marriage penalty rises, aggregate marriage rates fall. There is a greater impact on the timing of marriage, with couples often delaying marriage late in the year to minimise the marriage penalty. There is also some evidence that high marginal rates encourage divorce, especially on the part of women who are affected most by the marriage penalty (Whittington, Alm and Peters, 1990).
- 18 There is a considerable amount of economic research in the US on the effect of taxation on female labour supply; see Hausman (1979), Hayghe and Bianchi (1994), Eissa (1995a and b) and Eissa and Hoynes (2002). Edward McCaffery (2002 and 1997) discusses the major costs of entering the workforce as a US secondary earner and finds that while the incentives are quite different for lower- and middle-income families, both income groups are affected.
- 19 According to Apps and Rees (1999a), 'This issue is particularly important in OECD countries where the vast majority of workers earn around the median wage and the distribution is positively skewed. Under these conditions, tax systems, such as joint taxation (in Germany and the US) or EITC (in the US and proposed in the UK), have the effect of imposing higher rates on working married women and especially those married to low-earning husbands. Under system of joint income, both partners face the same marginal tax rate. However, what is missed is that two couples with same wages, non-labour incomes, demographics, and preferences

- face very different rates if there is heterogeneity in the labour supply of married women. Couples in which the female partner works typically face much higher marginal tax rates.'
- 20 Bargain, Laisney and Moreaux (2002) suggest the lack of simple econometric technique when accounting for participation decisions and tax-benefit system is the reason for the delays in the use of collective models for policy recommendation purposes. They also compare the predictions in term of policy analysis between the unitary and the collective models. The exercise consists in assessing a current French tax reform using a unitary model estimated on data generated by the collective one. The size of errors from the unitary model, both on positive and normative conclusions, suggests that more effort should be devoted to the estimation and operationalisation of collective household models with taxation.
- 21 There is often also a third, voluntary pillar for people who want more money for retirement in old age.
- 22 Most of the work on social security and gender focuses on coverage and benefits, as opposed to tax incidence and revenue.
- 23 See Apps and Rees (2002) for a discussion of some of these biases. On the other hand, in the US, Armour and Pitts (2002) find that the system favours women as they face a lower marginal rate at each stage. This tax differential, which is attributed to women's longer life expectancies, is seen as a desirable feature because of empirical evidence showing that female labour supply is more elastic than male labour supply.
- 24 Barrientos (1998) also finds the availability of personal pension plans would not raise the pension gender gap (which is almost non-existent in Chile) and may operate to close it.
- 25 VATs replaced sales and turnover taxes, which are seen to create a number of producer 'distortions'.
- 26 Ebrill *et al.* (2001) note that VATs in these countries show considerable diversity in the range of inputs for which tax offsetting is available and the range of economic activity (the base) to which the tax applies.
- 27 A recent theoretical paper by Emran and Stiglitz (2002) exploring selective indirect tax reform in developing countries argues that the existence of a large informal economy greatly reduces the desirability of VAT because it reduces welfare. This is the case when VAT reform involves both comprehensive rate increases, as well as when changes apply only to a subset of the commodities under the tax net.
- 28 We are indebted to Haroon Akram-Lodhi for this point.
- 29 Other issues are that the corporate tax alters three important business decisions regarding: a) legal form, i.e. whether to incorporate or to remain a privately-owned business or partnership (some countries also tax income of these business forms, but tax rate differentials generally remain, distorting some choices); b) equity financing vs. debt, since the latter reduces tax liability although it increases the risk of bankruptcy; c) dividend payments (taxable as income) and retained earnings (taxed as capital gains, generally at a lower rate).
- 30 This section draws heavily on the excellent paper by Palley (2001).
- 31 Personal correspondence with the authors.
- 32 Bird and Miller's (1989) study of the incidence of indirect taxes on low-income households in Jamaica is one of the few examples of a tax incidence analysis that incorporates gender variables. It divides the sample of low-income households into six expenditure groups representing low-income, medium-level low-income and higher lower-income. In addition, the analysis distinguishes coupled-headed and female-headed households and considers the age composition and dependency burdens within households.
- 33 CGE models are non-linear, multisectoral models which simulate the workings of a market economy. They have been used to analyse such issues as long-term growth and structural change, investment allocation, choice of development strategy, income distribution, trade policy and structural adjustment (see Robinson, 1989).

# **Bibliography**

- Aaberge, Rolf et al. (1989). Skatt, arbeidstilbud og inntektsfordeling (Taxation, Labour Supply and Income Distribution). Stockholm: Nordiska skattevetenskapliga forskningsrådets skriftserie, NSFS 22.
- Adams, Jolene (2001). 'Women and Social Security: How Far Are We?', IDASA Budget Brief No. 75. Cape Town and Pretoria, South Africa: IDASA (Institute for Democracy in South Africa). http://www.idasact.org.za/bis/
- Agarwal, Bina (1995). A Field of One's Own: Gender and Land Rights in South Asia. Cambridge, UK: Cambridge University Press.
- Alvarez, Michael and McCaffery, Edward (2000). 'Is There a Gender Gap in Fiscal Political Preferences?', Cal Tech Social Science Working Paper 1101. Pasadena, California: California Institute of Technology. http://www.hss.caltech.edu/CSLP/alvarez6.pdf
- (1998). 'Gender and Tax', Cal Tech Social Science Working Paper 1046. Pasadena, California: California Institute of Technology. http://www.hss.caltech.edu/SSWPLinks.html
- Anderson, Donna (1999). 'Tax Policy'. In Peterson, Janice and Margaret Lewi (eds), *The Elgar Companion to Feminist Economics*. Cheltenham, Glos, UK and Northampton, MA, USA: Edward Elgar Publishing, Inc.
- Anker, Richard (1998). Gender and Jobs: Sex Segregation of Occupations in the World. Geneva: International Labour Office.
- Apps, Patricia (2002a). 'Family-friendly policies make for higher fertility rates', Online Opinion website, posted 15 September 2002. http://www.onlineopinion.com/au/2002/Sep02/Apps.htm
- —— (2002b). 'Gender, Time Use and Models of the Household'. Sydney, Australia: University of Sydney.
- —— (2002c). 'Why an earned income tax credit program is a mistake for Australia', Australian Journal of Labour and Labour Economics 5(4): 549–568.
- Apps, Patricia and Rees, Ray (2002). 'Fertility, dependency and social security', Australian Journal of Labour Economics 5(4): 569–585.
- —— (1999a). 'Individual versus joint taxation in models with household production', The Journal of Political Economy 107(2): 393–403.
- —— (1999b). 'On the taxation of trade within and between households', *Journal of Public Economics* 73(2): 241–263.
- —— (1988). 'Taxation and the household', *Journal of Public Economics* 35: 155–169. Arenas de Mesa, Alberto and Bertranou, Fabio (1997). 'Learning from Social Security

- reforms: two different cases, Chile and Argentina', World Development 25(3): 329-48.
- Armour, Brian S. and Pitts, M. Melinda (2002). 'Incorporating Insurance Rate Estimates and Differential Mortality into Net Marginal Social Security Tax Rates Calculation', Working Paper 2002–29. Atlanta, GA: Federal Reserve Bank of Atlanta.
- Artecona, Raquel and Cunningham, Wendy (2002). 'Effects of Trade Liberalization on the Gender Wage Gap in Mexico', Policy Research Report Working Paper No. 21. Washington, DC: World Bank.
- Aslaksen, Julie and Koren, Charlotte (1992). 'Taxation, time use, and the value of unpaid labour: policy implications for the redistribution of income', *Review of Radical Political Economics* 24(2): 8–16.
- Atkinson, Anthony and Stiglitz, Joseph (1980). Lectures on Public Economics. New York: McGraw Hill Book Company.
- Bahl, Roy and Linn, Johannes (1992). *Urban Public Finance in Developing Countries*. New York: Oxford University Press.
- Bajtelsmit, Vickie and Bernasek, Alexandra (1996). 'Why do women invest differently than men?', Financial Counseling and Planning, Vol. 7: 1–10.
- Baker, Dean (2002). 'Tobin Taxes Are They Enforceable?'. Washington, DC: Center for Economic and Policy Research.
- Baker, Dean and Kar, Debayani (2002). 'The Privatization of Public Pension Systems in Developing Nations'. Washington, DC: Center for Economic and Policy Research.
- Barker, Drucilla K (1995). 'Economists, social reformers and prophets: a feminist critique of economic efficiency', *Feminist Economics* 1(3): 26–39.
- Bardhan, Pranab, Bowles, Samuel and Gintis, Herbert (1998). 'Wealth Inequality, Wealth Constraints and Economic Performance', Center for International and Development Economics Research (CIDER) Working Paper C98-097. Berkeley, CA: University of California at Berkeley.
- Bargain, Oliver, Laisney, François and Moreau, Nicolas (2002). 'Positive and Normative Analysis of Tax Policy: Does the Representation of the Household Decision Process Matter? Evidence for France'. Paris, Strasbourg and Mannheim, and Marseille: DELTA, BETA-THEME, Université Louis Pasteur and ZEW, and GREQAM.
- Barrientos, Armando (1998). 'Pension reform, personal pensions and gender differences in pension coverage', World Development 26(1): 125–137.
- Bartle, John and Rubin, Marilyn (2002). 'The Potential of Gender Budgeting: Has Its Day Come?', paper for panel on 'Democratizing the Budgetary Process' at Association for Budgeting and Financial Management Annual Conference, October 2002, Kansas City.
- Bartlett, Bruce (1998). The Marriage Penalty: Origins, Effects And Solutions, Tax Notes Special Report, 14 September, p. 1341
- Baughman, Reagan and Dickert-Conlin, Stacy (2003). 'Did expanding the EITC

- promote motherhood?', paper presented at the American Economic Association Annual Meeting, January 2003, Washington, DC.
- Bénabou, Roland (1996). 'Unequal Societies: Income Distribution and the Social Contract', New York C.V. Starr Center Working Paper No. 9617. New York: New York University.
- Bergmann, Barbara (1995). 'Becker's theory of the family: preposterous conclusions', Feminist Economics 1(1): 141–150.
- —— (1981). 'The economic risks of being a housewife', *The American Economic Review* 71(2): 81–86.
- Bernasek, Alexandra (2000). 'Gender, Risk and Investment: A Feminist Perspective', paper presented at the Annual IAFFE Conference, 15–17 August, Boğaziçi University, Istanbul, Turkey.
- Bird, Richard M. and Miller, Barbara Diane (1989). 'The Incidence of Indirect Taxes on Low-Income Households in Jamaica', *Economic Development and Cultural Change*. Chicago, IL: The University of Chicago.
- Bird, Richard M. and Zolt, Eric M. (2003). 'Introduction to Tax Policy Design and Development', paper prepared for a course on Practical Issues of Tax Policy in Developing Countries, 28 April–1 May, World Bank, Washington, DC.
- Birdsall, Nancy and Orivel, François (1996). 'Demand for primary schooling in rural Mali: should user fees be increased?', Education Economics 4(3): 279–296.
- Blundell, Richard, Duncan, Alan, McCrae, Julian and Meghir, Costas (2000). 'The Labour Market Impact of the Working Families Tax Credit'. London: Institute for Fiscal Studies. http://www.nottingham.ac.uk/~lezad/downloads/FS\_WFTC.pdf
- Blundell, Richard, Duncan, Alan and Meghir, Costas (1998). 'Estimating labour supply responses using tax policy reforms', *Econometrica* 66(4): 827–861.
- —— (1992). 'Taxation and empirical labour market supply models: lone parents in the UK', *Economic Journal* 102: 265–78.
- Blundell, Richard and Hoynes, Hilary (2001). 'Has "In-Work" Benefit Reform Helped the Labour Market?', NBER Working Paper w8546. Cambridge, MA: National Bureau of Economic Research. http://www.nber.org/books/bcf/benefit5-6-01.pdf
- Blundell, Richard and Reed, Howard (2000). 'The Employment Effects of the Working Families Tax Credit'. London: Institute for Fiscal Studies. http://www.ifs.org.uk/labmarket/wftc\_bn.pdf
- Bradford, David (1996). 'Consumption Taxes: Some Fundamental Transition Issues in Consumption Taxation'. Washington, DC: AEI Press.
- (1987). 'On the incidence of consumption taxes'. In Charles E. Walker and Mark A. Bloomfield (eds), *The Consumption Tax: A Better Alternative*. Cambridge, MA: Ballinger.
- —— (1982). 'The choice between income and consumption taxes', *Tax Notes* 16(8): 715–723.

- Bradshaw, Jonathan *et al.* (1996). 'Policy and the Employment of Lone Parents in 20 Countries: The EU Report'. European Observatory on National Family Policies, Social Policy Research Unit, York, UK: EU/University of York.
- Braunstein, Elissa and Folbre, Nancy (2001). 'To honor and obey: efficiency, inequality and patriarchal property rights', *Feminist Economics* 7(1):25–44.
- Brewer, Mike (2000). 'Comparing In-Work Benefits and Financial Work Incentives for Low-Income Families in the U.S. and U.K', IFS Working Paper W00/16. London: Institute for Fiscal Studies.
- Brown, Karen and Fellows, Mary Louise (eds) (1996). *Taxing America*. New York: New York University Press.
- Budget Watch (2001). 'Provincial taxes and equity: An impossible balance?', Issue 22. Budlender, Debbie (2000). 'The political economy of women's budgets in the south', World Development 28(7): 1365–1378.
- Budlender, Debbie (ed.) (2000). 'Introduction'. In *The Fifth Women's Budget*. Cape Town and Pretoria, South Africa: IDASA (Institute for Democracy in South Africa). http://www.internationalbudget.org/resources/library/5thwomenbudget.pdf http://idasact.org.za
- —— (1996). The Women's Budget. Cape Town and Pretoria, South Africa: IDASA (Institute for Democracy in South Africa).
- Budlender, Debbie and Hewitt, Guy (2002). Gender Budgets Make More Cents: Country Studies and Good Practices. London: Commonwealth Secretariat.
- Budlender, Debbie, Hewitt, Guy and Mukhopadhyay, Tanni (2002). Gender Budgets Make Cents. London: Commonwealth Secretariat.
- Budlender, Debbie, Sharp, Rhonda and Allen, Kerri (1998). How to do a Gender-Sensitive Budget Analysis: Contemporary Research and Analysis. Canberra, Australia and London: Australian Agency for International Development and Commonwealth Secretariat.
- Burgess, Robin and Stern, Nicholas (1993). 'Taxation and development', *Journal of Economic Literature* 31(2): 762–830. http://netec.mcc.ac.uk/WoPEc/data/Articles/aeajeclitv:31:y:1993:i:2:p:762-830.html
- Çağatay, Nilüfer (2003). 'Gender budgets and beyond: feminist fiscal policy in the context of globalization'. In Kerr, Joanna and Caroline Sweetman (eds), Gender and Development: Women Reinventing Globalization. Oxford, UK: Oxfam Publishing.
- Çağatay, Nilüfer, Elson, Diane and Grown, Caren (1995). 'Introduction: Gender, Adjustment, and Macroeconomics', World Development. Vol. 23, No. 11.
- Callan, Tim and Sutherland, Holly (1997). 'The impact of comparable policies in European countries: micro-simulation approaches', *European Economic Review* 41(3–5): 627–633.
- Castel, Paulette and Fox, Louise (1999). 'Gender Dimensions of Pension Reform in the Former Soviet Union'. Washington, DC: World Bank. http://wb.forumone.com/

- knowledge/chiefecon/conferen/papers/gender.pdf
- Chernick, Howard (2000). 'Yes! Consumption taxes are regressive', Challenge 43(5).
- Chu, Ke-young, Davoodi, Mamid and Gupta, Sanjeev (2000). 'Income Distribution and Tax and Government Spending Policies in Developing Countries', IMF Working Paper WP/00/62. Washington, DC: IMF.
- Commonwealth Secretariat, International Development Research Centre and the United Nations Development Fund for Women (2000). 'Improving the Effectiveness of Applying a Gender Analysis to Government Budgets', prepared for an Inter-Agency Workshop, 26–27 April, Marlborough House, London, England.
- Coopoo, Sikhander (2000). 'Women and local government revenue'. In Debbie Budlender (ed.), *The Fifth Women's Budget*. Cape Town and Pretoria, South Africa: IDASA (Institute for Democracy in South Africa). http://www.worldbank.org/wbi/publicfinance/documents/gender/Coopoo.pdf
- Cossa, Ricardo, Heckman, James L. and Lochner, Lance (1999). 'Wage Subsidies and Skill Formation: A Study of the Earned Income Tax Credit', mimeo. http://lily.src.uchicago.edu/econ350/EITC\_handout.pdf
- Cox Edwards, Alejandra (2001). 'Social Security Reform and Women's Pensions'. Washington, DC: Gendernet, World Bank. http://www.eldis.org/static/DOC8966.htm
- Cuevas, Alfredo (1990). 'Reforming Intergovernmental Fiscal Relations in Argentina', IMF Working Paper. Washington, DC: IMF.
- Daly, Mary (2000). The Gender Division of Welfare: The Impact of the British and German Welfare States. New York: Cambridge University Press.
- Dasgupta, Indraneel (2001). 'Gender biased redistribution and intra-household distribution', European Economic Review 45(9): 1711–1722.
- Davies, Hugh Brendan and Joshi, H. (1994). 'Sex, sharing and the distribution of income', *Journal of Social Policy* 23(3): 301–340.
- Deere, Carmen Diana and Leon, Magdalena (2003). 'The gender asset gap: land in Latin America', World Development 31(6): 925–947.
- Department of Constitutional Development (1998). 'Providing Affordable Basic Services: Tariff Healthcare and Targeting Subsidies-Towards an Indigence Policy for Local Government'. Pretoria, South Africa. http://www.dplg.gov.za/Documents/Publications/MunicipalityGuideline.htm
- Devarajan, Shantayanan and Hossain, Shaikh (1995). 'The Combined Incidence of Taxes and Public Expenditures in the Philippines', Policy Research Working Papers. Washington, DC: World Bank. http://bjcu.uca.edu.ni/biblioteca/debates/Devaraja.pdf
- de Villota, Paloma and Ferrari, Ignacio (2001). 'The Impact of the Tax/Benefit System on Women's Work'. Madrid: Universidad Complutense de Madrid.
- Dickert, Stacy and Houser, Scott (1998). 'Taxes and transfers: a new look at the marriage penalty', *National Tax Journal* 51(2): 175–217.

- Dickert, Stacy, Houser, Scott and Scholtz, John Karl (1995). 'The earned income tax credit and transfer program: a study of labour market and participation'. In James Poterba (ed.), *Tax Policy and the Economy*, Vol. 9. Cambridge, MA: MIT Press.
- Dilnot, Andrew and Duncan, Alan (1992). 'Lone mothers, family credit and paid work', Fiscal Studies 13: 1–21.
- Dilnot, Andrew and McCrae, J. (1999). 'Family Credit and the Working Families' Tax Credit', IFS Briefing Note No. 3. London: Institute for Fiscal Studies.
- Diokno, Benjamin (1996). 'Philippine tax system: present state, future directions', PAID! Advocacy Resource Magazine, October 1996.
- Doss, Cheryl (1996). 'Do Households Fully Share Risk? Evidence from Ghana', University of Minnesota Staff Paper 96–10. St. Paul, MN: University of Minnesota. http://netec.mcc.ac.uk/WoPEc/data/Papers/wopminnas9610.html
- Duclos, Jean-Yves, Makdissi, Paul and Wodon, Quentin (2002). 'Socially-Efficient Tax Reforms', Series Cahiers de recherche No. 0201. Quebec, Canada: Université Laval Département d'économique.
- Due, John (1988). *Indirect Taxation in Developing Countries*. Baltimore, MD: Johns Hopkins University Press.
- Duflo, Esther and Udry, Christopher (2001). 'Intra-household Resource Allocation in Cote d'Ivoire: Social Norms, Separate Accounts and Consumption Choices', unpublished draft. MIT and Yale University. http://econ-www.mit.edu/faculty/eduflo/files/papers/ivoire6.pdf
- Duvvury, Nata (1996). 'Excise Revenue of Indian States', presentation at the National Workshop on Anti-Liquor Movement, National Commission for Women, 18–22 March, New Delhi, India.
- Eardley, Tony (1996). 'Tax and Social Security Policies Toward Families: Australia Compared with Countries of the EU', paper presented to the Australian Family Research Conference, 27–29 November, Brisbane, Australia. http://www.aifs.org.au/institute/afrcpapers/eardley.html
- Ebrill, Liam, Keen, Michael, Bodin, J.P and Summers, Victoria (2001). *The Modern VAT*. Washington, DC: International Monetary Fund.
- Edwards, Sebastian and Edwards, Alejandra Cox (2000). 'Economic reforms and labour markets: policy issues and lessons from Chile', *Economic Policy* 30 (April): 183–227. http://www.nber.org/papers/w7646.pdf
- Eichler, Margaret (1997). Family Shifts: Families, Policies and Gender Equality. Toronto: Oxford University Press.
- Eissa, Nada (1999). 'Review of Taxing Women', Journal of Economic Literature 37(2).
- —— (1995a). 'Tax and transfer policy and female labour supply', National Tax Journal Annual Proceedings 1995.
- —— (1995b). 'Taxation and Labour Supply of Married Women: The Tax Reform Act

- of 1986 as a Natural Experiment', NBER Working Paper No. w5023. Cambridge, MA: National Bureau for Economic Research.
- Eissa, Nada and Hoynes, Hilary (2002). 'Taxes and the Labour Market Participation of Married Couples: The Earned Income Tax Credit', draft. Berkeley, CA: University of California, Berkeley. http://emlab.berkeley.edu/users/eissa/eitc\_ls.pdf
- —— (2000). 'Tax and Transfer Policy, and Family Formation: Marriage and Cohabitation.' Berkeley, CA: University of California, Berkeley.
- (1998). 'The Earned Income Tax Credit and the Labour Supply of Married Couples', NBER Working Paper 6856. Cambridge, MA: National Bureau of Economic Research.
- Eissa, Nada and Liebman, Jeffrey (1996). 'Labour supply response to the earned income tax credit', *Quarterly Journal of Economics* 111(2): 605–637.
- Elson, Diane (1999). 'Labour markets as gendered institutions: equality, efficiency and empowerment issues', World Development 27(3): 611–627.
- —— (1998). 'Integrating gender issues into national budgetary policies and procedures: some policy options', *Journal of International Development*, November–December (10): 929–941.
- (1996). 'Gender-Neutral, Gender-Blind, or Gender-Sensitive Budgets? Changing the Conceptual Framework to Include Women's Empowerment and the Economy of Care', Commonwealth Ministers Responsible for Women's Affairs, Fifth Meeting, Port of Spain, Trinidad, November 1996.
- Elwood, David (1999). 'The Impact of the Earned Income Tax Credit and Other Social Policy Changes on Work and Marriage in the United States', mimeo. Cambridge, MA: John F. Kennedy School of Government, Harvard University.
- Emran, M. Shahe and Stiglitz, Joseph E. (2002). 'On Selective Indirect Tax Reform in Developing Countries'. Palo Alto, CA, Washington, DC and New York, NY: Stanford University, George Washington University and Columbia University.
- Esam, Peter and Berthoud, Richard (1991). 'Independent Benefits for Men and Women'. London: Policy Studies Institute.
- Esim, Simel (2000). 'Impact of Government Budgets on Poverty and Gender Equality', paper written for the Inter-Agency Workshop on Improving the Effectiveness of Integrating Gender into Government Budgets, 26–27 April, Commonwealth Secretariat, Marlborough House, London.
- Even, William E. and Macpherson, David A. (1994). 'Gender Differences in Pensions', *Journal of Human Resources* 29(2): 555–587.
- Feenberg, Daniel and Rosen, Harvey (1995). 'Recent developments in the marriage tax', *National Tax Journal* 48(1): 91–101.
- Felix, David and Sau, Ranjit (1996). 'On the revenue potential of phasing in of the Tobin Tax'. In Haq, M. ul, Kaul, I. and Grunberg, I. (eds), *The Tobin Tax: Coping with Financial Volatility*. New York: Oxford University Press.

- Fjeldstad, Odd Helge (2001). 'Who levies what taxes?' *Budget Watch*, Issue 22. http://www.idasa.org.za/m\_main.php?view=2&art\_id=876
- Floro, Maria Sagrario and Seguino, Stephanie (2002). 'Gender Effects on Aggregate Saving', Policy Research Report on Gender and Development. Working Paper Series No. 23. Washington, DC: World Bank
- Findings (2003). 'Decentralization in Africa: emerging trends and progress', Report 229. Washington, DC: World Bank
- Folbre, Nancy (1997). 'Gender coalitions: extrafamily influences on intrafamily inequality'. In Haddad *et al.* (eds), *Intrahousehold Resource Allocation in Developing Countries:* Models, Methods, and Policy. Baltimore, MD and London: Johns Hopkins University Press for the International Food Policy Research Institute.
- —— (1995). 'Holding hands at midnight: the paradox of caring labour', Feminist Economics 1(1): 73–92.
- —— (1994). Who Pays for the Kids: Gender and the Structures of Constraint. London: Routledge.
- Fontana, Marzia (2002). 'Modeling the Effects of Trade on Women: The Case of Zambia', IDS Working Paper 155. Brighton: Institute of Development Studies, April.
- —— (2001). 'Modeling the Effects of Trade on Women: A Closer Look at Bangladesh', IDS Working Paper 139. Brighton: Institute of Development Studies, October.
- Fox, Louise (1997). 'Pension reform in the post-communist transition economies'. In *Transforming Post-Communist Political Economies*. Washington, DC: National Academy Press. http://www.nap.edu/html/transform/ch14.htm
- Fraser, Nancy (1997). 'After the family wage: a postindustrial thought experiment'. In *Justice Interruptus*. New York and London: Routledge.
- Fullerton, Don and Rogers, Diane Lim (1993). Who Bears the Lifetime Tax Burden? Washington, DC: The Brookings Institution.
- Fultz, Elaine, Ruck, Markus and Steinhilber, Silke (eds) (2003). The Gender Dimensions of Social Security Reform in Central and Eastern Europe: Case Studies of the Czech Republic, Hungary and Poland. Budapest: International Labour Organisation. http://www.ilo-ceet.hu/download/gender.pdf
- Gauthier, Anne and Hatzius, Jan (1997). 'Family benefits and fertility: an econometric analysis', *Population Studies* 51(3): 295–306.
- Gertler, Paul and Glewwe, Paul (1992). 'The willingness to pay for education for daughters in contrast to sons: evidence from rural Peru', *The World Bank Economic Review* 6(1).
- Gertler, Paul, Locay, Luis and Sanderson, Warren. 'Are User Fees Regressive? The Welfare Implications of Health Care Financing Proposals in Peru', NBER Working Papers 2299. Cambridge, MA: National Bureau of Economic Research.
- Gillis, Malcolm (ed.) (1989). Tax Reform in Developing Countries. Durham, NC: Duke University Press.

- Gilson, Lucy (1996). 'The lessons of user fee experience in Africa'. In A. Beattie, J. Doherty, L. Gilson, E. Lambo and P. Shaw (eds), Sustainable Health Care Financing in Southern Africa, Papers from an EDI Health Policy Seminar held in June 1996 in Johannesburg, South Africa. http://www.worldbank.org/healthreform/library/sa/shaw3.pdf
- Gilson, Lucy, Russell, Steven and Buse, Kent (1995). 'The political economy of user fees with targeting: developing equitable health financing policy', *Journal of International Development* 7(3): 369–401.
- Ginn, Jay and Arber, Sara (1993). 'Pension penalties: the gendered division of occupational welfare', Work, Employment and Society 7(1): 47–70.
- Goldman, Tanya (2000). 'Customs and excise'. In Debbie Budlender (ed.), *The Fifth Women's Budget*. Cape Town and Pretoria, South Africa: IDASA. http://www.idasa.org.za/pdf/940.pdf
- Goode, Jackie, Callender, Claire and Lister, Ruth (1998). 'Purse or Wallet? Gender Inequalities and Income Distribution within Families on Benefits'. London: Policy Studies Institute.
- Gornick, Janet (1999). 'Gender Equality in the Labour Market: Women's Employment and Earnings', Luxembourg Income Study Working Paper No. 206. Maxwell School of Citizenship and Public Affairs. Syracuse, NY: University of Syracuse. http://www.lisproject.org/publications/liswps/206.pdf
- Greenhalgh, Susan (1985). 'Sexual stratification: the other side of growth with 'equity' in East Asia', *Population and Development Review* 1(2): 265–314.
- Gregg, Paul, Johnson, Paul and Reed, Howard (1999). 'Entering Work and the British Tax and Benefit System', IFS monograph. London: Institute for Fiscal Studies.
- Grown, Caren, Elson, Diane and Çağatay, Nilüfer (2000). 'Introduction: Special Issue on Growth, Trade, Finance and Gender Inequality', World Development 28(7).
- Grunberg, Isabelle (1998). 'Double jeopardy: Globalization, liberalization and the fiscal squeeze', World Development 26(4): 591–605.
- Gustafsson, Siv (1992). 'Separate taxation and married women's labour supply: a comparison of West Germany and Sweden', *Journal of Population Economics* 5: 61–85.
- Guyer, Jane (1988). 'Dynamic approaches to domestic budgeting: cases and methods from Africa'. In Daisy Dwyer and Judith Bruce (eds), A Home Divided: Women and Income in the Third World. Palo Alto, CA: Stanford University Press.
- Haddad, Lawrence, Hoddinott, J. and Alderman, H. (eds) (1997). *Intrahousehold Resource Allocation in Developing Countries: Methods, Models and Policy.* Baltimore, MD: The Johns Hopkins University Press.
- Hanfstaengl, Eva Maria (1999). 'Expert meeting on "taxing excessive financial speculation", CIDSE Advocacy Newsletter, Issue 9. http://www.cidse.be/en/advocacy/newsletter/an9tg2.htm

- Hanson, Sandra and Ooms, Theodora (1991). 'The economic costs and rewards of two-earner, two parent families', *Journal of Marriage and the Family* 53(3): 622–34.
- Haq, Mahbul, Kaul, Inge and Grunberg, Isabelle (eds) (1996). *The Tobin Tax:* Coping with Financial Volatility. Oxford, UK: Oxford University Press.
- Harberger, Arnold C. (2003). 'Reflections on Distributional Considerations and the Public Finances', paper prepared for a course on Practical Issues of Tax Policy in Developing Countries, 28 April–1 May, World Bank, Washington, DC.
- Harmelink, Philip J. and Speyrer, Janet Furman (1992). 'Alternative methods of taxing Social Security', *Journal of Post Keynesian Economics* 15(1): 3–30.
- Hartzenberg, Trudi (1996). 'Taxation'. In Debbie Budlender (ed.), *The Women's Budget*. Cape Town and Pretoria, South Africa: IDASA (Institute for Democracy in South Africa).
- Hausman, Jerry (1979). 'The effects of wages, taxes and fixed costs of women's labour force participation', *Journal of Political Economy* 14: 161–194.
- Hayghe, Howard and Bianchi, Suzanne (1994). 'Married mothers' work patterns: the job-family compromise', *Monthly Labour Review June*: 24–30.
- Himmelweit, Susan (2001). 'Tools for Budget Impact Analysis: Taxes and Benefits', paper prepared for UNIFEM-OECD-Nordic Council-Government of Belgium High Level Conference, 16–17 October, Brussels. http://www.worldbank.org/wbi/public finance/documents/gender/himmelweit.pdf
- —— (1999). 'Domestic Labour'. In Peterson, Janice and Margaret Lewis (eds), *The Elgar Companion to Feminist Economics*. Cheltenham, Glos., UK and Northampton, MA, USA: Edward Elgar Publishing, Inc.
- —— (1998). 'The need for gender impact analysis'. In S. Robinson (ed.), *The Purse or the Wallet?* London: Women's Budget Group.
- Hochschild, Adam (1998). King Leopold's Ghost. New York: Houghton Mifflin.
- Hofbauer Balmori, Helena (2003). 'Gender and Budgets. Overview Report'. Brighton, UK: BRIDGE, Institute of Development Studies, University of Sussex. http://www.ids.ac.uk/bridge/reports/CEP-Budgets-report.pdf
- Holtz-Eakin, Douglas and Poterba, James (eds) (2000). *The National Tax Journal*. Special Issue: The Economic Effects of Taxation, Vol. LIII, No. 3, Part 2.
- Hussain, Athar (2001). 'Pension Reform in China: Its Progress and Challenges', paper presented for Financial Sector Reform in China conference, September 2001.
- http://www.ksg.harvard.edu/cbg/Conferences/financial\_sector/Pension ReforminChina.pdf
- International Labour Organization (2002). Women and Men in the Informal Economy: A Statistical Picture. Geneva: ILO Employment Sector.
- International Monetary Fund (2002). Government Finance Statistics Yearbook. Washington, DC: IMF.
- James, Estelle (1998). 'Social Security Reform: Lessons from Other Nations', speech

- delivered to Heritage Foundation, April 1998, Washington, DC. http://www.estelle-james.com/downloads/heritage.pdf
- James, Estelle, Edwards, Alejandra Cox and Wong, Rebecca (2002). 'The Gender Impact of Pension Reform: A Cross-Country Analysis', World Bank Working Paper. Washington, DC: World Bank. http://www.estelle(james.com/downloads/gender.pdf
- Jepsen, Maria, Meulders, D., Plaman, O. and Vanhuynegem, P. (1997). 'Gender Inequalities in European Unemployment Benefit Systems: Individualization of the Fiscal and Social Rights and the Equal Opportunities Between Women and Men', Brussels: Département d'Économie Appliquée, Économie du Travail et de l'Emploi (ETE).
- Johnson, Nichola (1999). A Hand Up: How State Earned Income Tax Credits Help Working Families Escape Poverty (1999 ed.). Washington, DC: Center on Budget and Policy Priorities.
- Kesner-Skreb, Marina and Kulis, Danijela (2002). *The Citizen's Guide to Taxation*. Zagreb, Croatia: Institute of Public Finance.
- Khattry, Barsha and Rao, J. Mohan (2002). 'Fiscal faux pas? An analysis of the revenue implications of trade liberalization', *World Development* 30(8): 1431–1444.
- Killingsworth, Mark and Heckman, James (1986). 'Female labour supply: a survey'. In O. Ashenfelter and R. Layard (eds), *Handbook of Labour Economics* 1(2): 103–204.
- Kipp, Walter, Kamugisha, Jimmy, Jacobs, Phil, Burnham, Gilbert and Rubaale, Tom (2001). 'User fees, health staff incentives, and service utilization in Kabarole District, Uganda', World Health Organization Bulletin 79(11): 1032–7. Geneva.
- http://www.who.int/docstore/bulletin/pdf/2001/issue11/bu0969.pdf
- Kornhauser, Marjorie E. (1987). 'The rhetoric of the anti-progressive income tax movement: a typical male reaction', *Michigan Law Review* 86: 465–523.
- Le, Tuan Minh (2003). 'Value-Added Taxation: Mechanism, Design, and Policy Issues', paper prepared for the course on Practical Issues of Tax Policy in Developing Countries, 28 April–1 May, World Bank, Washington, DC
- Leuthold, Jane (1984). 'Income splitting and women's labour-force participation', *Industrial and Labour Relations Review* 38(1): 98–105.
- Lundberg, Shelly J. and Ward-Batts, Jennifer (2000). 'Saving for Retirement: Household Bargaining and Household Net Worth', Working Paper WP 2000–004. Ann Arbor, MI: University of Michigan Retirement Research Center.
- Lundberg, Shelly J., Pollak, Robert A. and Wales, T.J. (1997). 'Do husbands and wives pool their resources? Evidence from the UK child benefit', *Journal of Human Resources* 32(3): 462–80.
- Manser, Marilyn and Brown, Murray (1980). 'Marriage and household decision making', *International Economics Review* 21(1): 31–44.
- Marshall, Richard (2003). 'The Gender Dimensions of Direct Taxes in Mongolia'. New York: UNDP.
- Martinez-Vasquez, Jorge and McNab, Robert M. (2003). 'Fiscal decentralization and

- economic growth', World Development 31(9): 1597–1616.
- Mbugua, James, Bloom, G.H. and Segall, M.M. (1995). 'Impact of user charges on vulnerable groups: the case of Kibwezi in rural Kenya', Social Science and Medicine 41(6): 829–35.
- McCaffery, Edward (2002a). Fair Not Flat: How to Make the Tax System Better and Simpler. Chicago, IL: University of Chicago Press.
- —— (2002b). 'Women and Taxes', NCPA Policy Report No. 250. Dallas, TX and Washington, DC: The National Center for Policy Analysis. http://www.ncpa.org/ pub/st/st250/st250.pdf
- —— (1999). 'The burdens of benefits', Villanova Law Review 44(445).
- —— (1977). Taxing Women. Chicago: University of Chicago Press.
- McClure, Charles (2001). 'Decentralisation and individual income tax', Budget Watch Issue 22.
- Mehra, Rekha and Gammage, Sarah (1999). 'Trends, countertrends, and gaps in women's employment', World Development 27(3): 533–550.
- Meyer, Bruce and Rosenbaum, Dan (2000). 'Making Single Mothers Work: Recent Tax and Welfare Policy and its Effects', NBER Working Paper 7491. Cambridge, MA: National Bureau of Economic Research.
- Mitrusi, Andrew and Poterba, James (2000). 'The distribution of payroll and income tax burdens, 1979–1999', *National Tax Journal* 53: 765–794.
- Munnell, Alicia (1980). 'The couple versus the individual under the federal personal income tax'. In Henry J. Aaron and Michael J. Boskin (eds), *The Economics of Taxation*. Washington, DC: The Brookings Institution.
- Myck, Michael (2000). 'Distributional Consequences of Fiscal Reform since 1997', IFS Briefing Notes No. 14. London: Institute for Fiscal Studies. http://www.ifs.org.uk/taxben/fiscalreform.pdf
- Nanda, Priya (2002). 'Gender dimensions of user fees: implications for women's utilization of health care', *Reproductive Health Matters* 10(20): 127–134.
- National Center for Policy Analysis Daily Policy Digest (2002). 'Women and taxes', 5 March.
- Nelson, Julie (1996). 'Feminist theory and the income tax'. In Feminism, Objectivity, and Economics. New York: Routledge.
- Nelson, Julie (1991). 'Tax reform and feminist theory in the United States context: incorporating the human connection', *Journal of Economic Studies* 18(5/6): 11–29.
- Newton, L., Orr, L., Valodia, I. and Budlender, D. (1999). 'The budget and employment creation for women and men'. In Debbie Budlender (ed.) *The Fourth Women's Budget*. Cape Town and Pretoria: IDASA (Institute for Democracy in South Africa).
- New York Times (2000). 'Economic Scene: Tax System Discourages Married Women From Working', 2 November 2000, Section C, Page 2, Column 1. http://www.nytimes.com/2000/11/02/business/02SCEN.html

- O'Donoghue, Cathal and Sutherland, Holly (1999). 'Accounting for the family in European income tax systems', Cambridge Journal of Economics 23: 565–598.
- OECD (Organization for Economic Cooperation and Development) (2001). Consumption Tax Trends: VAT/GST, Excise and Environmental Taxes 2001. Paris. http://www1.oecd.org/publications/e-book/2301111e.pdf
- —— (1993). Taxation in OECD Countries. Paris.
- O'Neill, June (1983). 'Family issues in taxation'. In Rudolph G. Penner (ed.), *Taxing the Family*. Washington, DC: American Enterprise Institute for Public Policy Research.
- Ono, Hiromi and Stafford, Frank (2001). 'Till death do us part or I get my pension? Wives' pension holding and marital dissolution in the United States', Scandinavian Journal of Economics 103(1): 1–20.
- Oostendorp, Remco (2002). 'Does Globalization Reduce the Gender Wage Gap?', mimeo. Amsterdam, The Netherlands: Economic and Social Institute, Free University of Amsterdam.
- Ott, Katarina (ed.) (2000). A Citizen's Guide to the Budget. Zagreb, Croatia: Institute of Public Finance.
- Palley, Thomas (2001). 'Destabilizing Speculation and the Case for an International Currency Transactions Tax', Challenge May—June 2001, 44(3): 70–89.
- Palmer, Ingrid (1995). 'Public finance from a gender perspective', World Development 23(11): 1981–86.
- Palsson, Anne-Marie (1997). 'Taxation and the market for domestic services'. In Christina Jonung and Inga Persson (eds), Economics of the Family and Family Policies. London: Routledge.
- Pechman, Joseph and Engelhardt, Gary (1990). 'The income tax treatment of the family: an international perspective', *National Tax Journal* 43(1): 1–32.
- Philipps, Lisa (2002). 'Tax Law and Social Reproduction: The Gender of Fiscal Policy in an Age of Privatization'. In B. Cossman and J. Fudge (eds), *Privatization*, *Law and the Challenge to Feminism*. Toronto: University of Toronto Press.
- Philipps, Lisa (1996). 'Tax Policy and the Gendered Distribution of Wealth'. In Bakker, Isabella, *Rethinking Restructuring: Gender and Change in Canada*. Toronto: University of Toronto Press.
- Phipps, Shelley and Burton, Peter (1998). 'What's mine is yours? The influence of male and female incomes on patterns of household expenditure', *Economica* 65: 599–613.
- Piggott, John and Whalley, John (1996). 'The tax unit and household production', *Journal of Political Economy* 104(2): 398–418.
- Pollin, Robert, Baker, D., and Schaberg, M. (1999). 'Securities Transactions Tax for U.S. Financial Markets', paper presented for the conference in honor of Hyman P. Minsky, 21–23 April, Jerome Levy Institute, Blithewood, NY.
- Rake, Katherine and Daly, Mary (1999). 'Gender, Household and Individual Income

- in France, Germany, Italy, the Netherlands, Sweden , the USA, and the UK', Luxembourg Income Study Working Paper No. 332. Maxwell School of Citizenship and Public Affairs. Syracuse, NY: University of Syracuse. http://www.lisproject.org/publications/liswps/332.pdf
- Rao, J. Mohan (1999). 'Globalization and the Fiscal Autonomy of the State', Human Development Report Background Papers 1999, Vol. 1. New York: UNDP.
- Reddy, Sanjay and Vandemoortele, Jan (1996). 'User Financing of Basic Social Services: A review of theoretical arguments and empirical evidence', UNICEF Staff Working Papers. Evaluation, Policy and Planning Series. New York: UNICEF.
- Reeves, Hazel and Wach, Heike (1999). 'Women's and gender budgets: an annotated resource list', BRIDGE Bibliography No. 9. Brighton, UK: BRIDGE, Institute of Development Studies, University of Sussex.
- Reiss, Wolfram (2000). 'Value Added Taxation in Germany', Bulletin for International Fiscal Documentation 54(8–9): 405–413.
- Robinson, S. (1989). 'Multisectoral Models'. In Chenery, H. and Srinivasan, T.N. (eds), *Handbook of Development Economics*, *Volume II*, , pp. 885–947. Elsevier Science Publishers, B.V.
- Rochjadi, Achmad and Leuthold, Jane H. (1994). 'The Effect of Taxation on Labour Supply in a Developing Country: Evidence from Cross-sectional Data'. Chicago, IL: The University of Chicago.
- Rodrik, Dani (1997). 'Trade, Social Insurance and the Limits to Globalization', NBER Working Paper 5905. Cambridge, MA: NBER (National Bureau of Economic Research).
- Rosen, Harvey S. (2002). Public Finance (sixth ed.). New York: McGraw Hill.
- Rusimbi, Mary (n.d.). A Case Study on Strategies for Transforming and Mainstreaming Gender into National Planning and Budgeting Process and Macro-Economic Policies in Tanzania. Kampala: Tanzania Gender Networking Programme.
- Sainsbury, Diane (1996). Gender, Equality and Welfare States. Cambridge, UK: Cambridge University Press.
- Sandmo, Agnar (1990). 'Tax distortions and household production', Oxford Economic Papers 42(1): 78–90.
- Sandford, Cedric (2000). Why Tax Systems Differ: A Comparative Study of the Political Economy of Taxation. Bath, UK: Fiscal Publications.
- Sarraf, Feridoun (2003). 'Gender-Responsive Government Budgeting', IMF Working Paper WP/03/83. Washington, DC: IMF.
- Schmidt, Rodney (1999). 'A Feasible Foreign Exchange Transactions Tax'. Ottawa, Canada: The North-South Institute. http://www.networkideas.org/alt/mar2002/alt15 Tobix Tax.htm
- Schmitt, John (2003). 'Is It Time to Export the US Tax Model to Latin America?', Center for Economic and Policy Research Issue Brief, March. Washington, DC:

- CEPR. www.cepr.net/Latin\_American\_taxes.htm
- Seguino, Stephanie (2000). 'Accounting for Asian economic growth: Adding gender to the equation', *Feminist Economics* 6(3): 22–58.
- Sen, Amaryta (1990). 'Gender and cooperative conflicts'. In Irene Tinker (ed.), Persistent Inequalities: Women and World Development. New York: Oxford University Press.
- Sen, Gita (2000). Gender Mainstreaming in Finance. A Reference Manual for Governments and Other Stakeholders. London: Commonwealth Secretariat.
- Shapiro, Daniel (1999). 'Effective Marginal Tax Rates on Low-Income Households'. Washington, DC: Employment Policies Institute.
- Shapiro, Isaac (ed.) (2001). A Guide to Budget Work for NGOs. Washington, DC: The International Budget Project.
- Shaw, Lois and Hill, Catherine (2002). 'The Gender Gap in Pension Coverage: What Does the Future Hold?', Washington, DC: IWPR (Institute for Women's Policy Research).
- Shepard, Donald, Carrin, Guy and Nyandagazi, Prosper (1992). 'The role of user fees in financing health care at government health centres in Rwanda'. In Carrin, Guy (ed.), Strategies for Health Care Finance in Developing Countries. New York: St. Martin's Press.
- Shibata, Aiko (1994). 'The effects of Japanese income tax provisions on women's labour force participation'. In Folbre, Nancy, Bergmann, B., Agarwal, B. and Floro, M. (eds), *Women's Work in the World Economy*. New York: New York University Press.
- Shome, Parthasarathi (1999). 'Taxation in Latin America: Structural Trends and Impact of Administration', IMF Working Paper. Washington, DC: IMF. http://www.imf.org/external/pubs/ft/wp/1999/wp9919.pdf
- Shome, Parthasarathi (ed.) (1995). *Tax Policy Handbook*. Washington, DC: International Monetary Fund.
- Smith, Terence (2000). 'Women and tax in South Africa'. In Debbie Budlender (ed.) *The Fifth Women's Budget*. Cape Town and Pretoria, South Africa: IDASA (Institute for Democracy in South Africa). http://www.worldbank.org/wbi/publicfinance/documents/gender/smith.pdf
- Standing, Guy (1999). 'Global feminization revisited', World Development 27(7): 583–602.
- —— (1989). 'Global feminization through flexible labour', World Development 17(7): 1077–95.
- Staudt, Nancy (1996). 'Taxing housework', Georgetown Law Journal 84: 1571–1647.
- Stiglitz, Joseph E. (2000). *Economics of the Public Sector* (third ed.). New York: W.W. Norton.
- Stotsky, Janet (1997a). 'Gender bias in tax systems', *Tax Notes International*, 9 June, pp. 1913–1923.

- Stotsky, Janet (1997b). 'How tax systems treat men and women differently', *Finance and Development*, March: 30–33.
- Stotsky, Janet and WoldeMariam, Asegedech (2002). 'Central American Tax Reform: Trends and Possibilities', IMF Working Paper WP/02/227, December. Washington, DC: IMF.
- Strøm, Steiner, Aaberge, R., Colombino, U. and Wennemo, T. (2000). 'Joint labour supply of married couples: efficiency and distribution effects of tax and labour market reforms'. In Mitton, L., Sutherland, H. and Weeks, M. (eds), Microsimulation Modelling for Policy Analysis. Challenges and Innovations. Cambridge, UK: Cambridge University Press.
- Sunden, Annika and Surette, Brian (1998). 'Gender Differences in the Allocation of Assets in Retirement Savings Plans', American Economic Review: 207–211.
- Sutherland, Holly (1999). 'EUROMOD: A European Benefit Tax Model and Social Integration', Preparatory Study Final Report to the European Commission.
- —— (1997). 'Women, men and the redistribution of income', *Fiscal Studies* 18(1): 1–22. http://www.ifs.org.uk/publications/fiscalstudies/fsholly.pdf
- Tanzi, Vito (1996). 'Globalization, Tax Competition and the Future of Tax Systems', IMF Fiscal Affairs Working Paper WP/96/141. Washington, DC: IMF.
- Tax Notes (2002). Men, Women, Taxes, and Ethics (Special Reports), 21 October, p. 401. Tax Notes (1997). Privatization Of Social Security A Threat To Women (Viewpoint), 5 May, p. 717.
- Tax Notes (1994). Tax Policy, Gender Equity: Women's Issues in Rural Areas of New Zealand, (Tax Article Abstracts), 5 September, p. 1340.
- Tax Notes International (1997). Gender Bias In Tax Systems (Opinion and Analysis: Special Report), 9 June, p. 1913.
- Thirsk, Wayne (ed.) (1997). Overview: The Substance and Process of Tax Reform in Eight Developing Countries. Washington, DC: World Bank.
- Thomas, Duncan (1992). 'The Distribution of Income and Expenditure within the Household', prepared for IFPRI World Bank Conference on Intrahousehold Resource Allocation: Policy Issues and Research Methods, 12–14 February, IFPRI, Washington, DC.
- Townson, Monica (1997). 'Women and the Economy: Long-Term Policy Research Issues.' Ottawa, Ontario: Status of Women Canada. http://www.swc-cfc.gc.ca/pubs/womeneconomy/womeneconomy\_e.html
- Tzannatos, Zafiris (1999). 'Women and labour market changes in the global economy: growth helps, inequalities hurt and public policy matters', *World Development* 27(3): 551–569.
- United Nations (2000). The World's Women 2000. Trends and Statistics. New York.
- United Nations Development Programme (1995). *Human Development Report* 1995. New York and Oxford: Oxford University Press.

- United States Government (2000). Budget of the US Government 2000, Historical Tables, Table 2.2. Washington, DC: U.S. Government Printing Office.
- Vandemoortele, Jan (2001). 'Are User Fees and Narrow Targeting Gender-Neutral?' New York: UNDP.
- —— (2000). 'Absorbing Social Shocks, Protecting Children and Reducing Poverty', UNICEF Paper 00–001. New York: United Nations Children's Fund (UNICEF).
- Van Staveren, Irene and Akram-Lodhi, A. Haroon (2003). 'A Gender Analysis of the Impact of Indirect Taxes on Small and Medium Enterprises in Vietnam.' The Hague, The Netherlands: Institute of Social Studies. Draft paper presented at the IAFFE Conference, 27–29 June, University of the West Indies, Barbados.
- Vella, Francis (1993). 'Nonwage benefits in a simultaneous model of wages and hours: labour supply functions of young females', *Journal of Labour Economics* 11(4): 704–723.
- Vermeulen, Frederic (2003). 'A Collective Model for Female Labour Supply with Nonparticipation and Taxation'. Leuven, Belgium: Center for Economic Studies, University of Leuven.
- Walker, Robert and Wiseman, Michael (1997). 'The possibility of a British earned tax credit', Fiscal Studies 18(4): 401–425. http://www.ifs.org.uk/publications/fiscalstudies/fswalwis.pdf
- Whittington, Leslie, Alm, James and Peters, Elizabeth (1990). 'Fertility and the personal exemption: implicit pronatalist policy in the United States', *American Economic Review* 80(3): 545–556.
- Witt, Daniel A. and McClure, Charles E. (2002). 'Tax Reform: Creating the Modern System'. Washington, DC: International Tax and Investment Center. http://ia.ita.doc.gov/download/kazakhstan-nme-status/itic/itic-comments-kaz.pdf
- Woolley, Frances (1998). 'For a radical redesign of our tax treatment of the family', *Policy Options*, September 1998, 7–10. http://www.irpp.org/po/archive/sep98/woolley. pdf
- World Bank (1991). Lessons of Tax Reform. Washington, DC.
- World Bank (2003). World Development Report 2004: Making Services Work for Poor People. Washington, DC: World Bank Publications.
- Young, Claire (2000). 'Women, Tax and Social Programs: The Gendered Impact of Funding Social Programs Through the Tax System'. Ottawa, Ontario: Status of Women Canada. http://www.swc-cfc.gc.ca/pubs/factsheets/200211\_020510\_42\_e.html

This title was commissioned by the Commonwealth Secretariat as part of its commitment to integrate gender concerns into economic policy. It provides information to assist in the analysis of potential gender bias in tax systems and the design of gender-sensitive revenue measures.

£10.00



Commonwealth Secretariat

